GREEN TARIFF SHARED RENEWABLES (GTSR) COMMUNITY RENEWABLES (CR) PROGRAM PROJECT DEVELOPMENT RIDER AND AMENDMENT

to the

RENEWABLE POWER PURCHASE AGREEMENT

hetween

SOUTHERN CALIFORNIA EDISON COMPANY

and

[NAME OF SELLER]

This Green Tariff Shared Renewables ("GTSR") Community Renewables ("CR") Program Project Development Rider and Amendment ("GTSR CR Rider") to the Agreement (as that term is defined below) dated as of the GTSR CR Rider Effective Date (as that term is defined below) is entered into between Southern California Edison Company, a California corporation ("SCE"), and [Name of Seller], a [Legal Status of Seller] ("Seller"). SCE and Seller are hereinafter sometimes referred to individually as a "Party" and jointly as the "Parties". Capitalized terms used herein and not otherwise defined in this GTSR CR Rider shall have the meanings ascribed to such terms in the Agreement.

RECITALS

The Parties enter into this GTSR CR Rider with reference to the following facts:

- A. Concurrently herewith, SCE and Seller enter into that certain Renewable Power Purchase Agreement, (as amended from time to time, the "Agreement"), under which, among other things, Seller is willing to construct, own, and Operate a Generating Facility, consisting of Distributed Energy Resource(s), each of which qualifies, or will qualify, as an ERR, and to sell the Product to SCE, and SCE is willing to purchase the Product from Seller, pursuant to the terms and conditions set forth in this Agreement.
- B. The Parties seek to modify the Agreement in order to incorporate provisions related to the GTSR-CR Decisions.

AGREEMENT

In consideration of the promises, mutual covenants and agreements hereinafter set forth, and for other good and valuable consideration, as set forth herein, the Parties agree to amend the Agreement as follows:

- 1. The <u>List of Exhibits</u> is amended to add the following at the end thereof:
 - "P. Subscribed Customer Reporting Form.
 - Q-1. QF Efficiency Monitoring Program Cogeneration Data Reporting Form.
 - Q-2. Fuel Use Standards Small Power Producer Data Reporting Form.
 - R-1. Community Interest Requirement Attestation Form.
 - R-2. Expression of Interest Form.
 - R-3. Commitment to Enroll Form."
- 2. Section 1.01 is deleted in its entirety and replaced with the following:
 - "1.01 Generating Facility and Distributed Energy Resources.
 - (a) Name: [Generating Facility name and the name of each DER that constitutes the Generating Facility].
 - (b) Location of each DER Site:

[DER Site 1 Address:

DER Site 2 Address:

etc...,

each of which is further described in Exhibit B.]

{SCE Comment: Add a DER Site address for each individual DER. List as many DER Site addresses as there are actual DERs}

- (c) Description: As set forth in Exhibit B.
- (d) Product: All electric energy produced by the Generating Facility throughout the Delivery Term, net of Station Use; [and Site Host Load] {SCE Comment: For DERs with Excess-Sales only}, all Green Attributes [associated with Qualified Amounts] {SCE Comment: For DERs with Excess-Sales only}; all Capacity Attributes; and all Resource Adequacy Benefits; generated by, associated with or attributable to the Generating Facility throughout the Delivery Term.
- (e) Interconnection Point: [insert the names or locations of each DER and identify the PNode associated with each DER].

- {SCE Comment: Placeholder for name of substation or method of identifying the location of interconnection to Transmission Provider's electric system. First point of interconnection must be within SCE's service territory.}
- (f) Delivery Point: At the point of interconnection with the CAISO-Controlled Grid, [insert name or location].

 {SCE Comment: Placeholder for identifying location on the CAISO-Controlled Grid. For DERs, this will be the location on the CAISO-Controlled Grid associated with the Distributed Energy Resource Aggregation}.
- (g) ERR Type: [Generation Technology].
- (h) Contract Capacity and DER Contract Capacity:
 - (i) Contract Capacity: [Number] MW. {SCE Comment: This should equal the total AC nameplate capacity of the Generating Facility. For DERs, this will equal sum of all DER Contract Capacities.} The Contract Capacity may be reduced as set forth in Section 3.06(a).
 - (ii) DER Contract Capacity:

[DER Contract Capacity 1: [Number] MW;

DER Contract Capacity 2: [Number] MW; etc...]

{SCE Comment: Add a DER Contract Capacity for each individual DER. List as many DER Contract Capacities as there are actual DERs, and each should equal the AC nameplate capacity for each individual DER.}

Each DER Contract Capacity may be reduced as set forth in Section 3.06(a).

- (i) Installed DC Rating and DER Installed DC Rating:
 - (i) Installed DC Rating: [Number] kW_{PDC}. {SCE Comment: This should equal the Installed DC Rating of the Generating Facility. For DERs, this will equal sum of all DER Installed DC Ratings.} The Installed DC Rating may be reduced as set forth in Section 3.06(b). {SCE Comment: For Solar Photovoltaic.}
 - (ii) DER Installed DC Rating:

[DER Installed DC Rating 1: [Number] kW_{PDC}.

[DER Installed DC Rating 1: [Number] kW_{PDC}; etc...]

{SCE Comment: Add a DER Installed DC Rating for each individual DER. List as many DER Installed DC Ratings as there are actual DERs, and each should equal the Installed DC Rating for each individual DER}

Each DER Installed DC Rating may be reduced as set forth in Section 3.06(b).

{SCE Comment: For Solar Photovoltaic DERs only}

(j) Expected Annual Net Energy Production. {SCE Comment: For all technologies except Solar Photovoltaic.}

The Expected Annual Net Energy Production for each Term Year will be the value calculated in accordance with the following formula:

EXPECTED ANNUAL NET ENERGY PRODUCTION, in kWh

= A x B x C [- D] {SCE Comment: For DERs with Excess-Sales only}

Where:

A = Contract Capacity in kW.

B = [Number] % capacity factor.

C = 8,760 hours per year.

[D = Expected Annual Site Host Load kWh per Term Year] {SCE Comment: For DERs with Excess-Sales only}

Expected Annual Net Energy Production. {SCE Comment: For Solar Photovoltaic.}

The Expected Annual Net Energy Production for each Term Year will be the value calculated in accordance with the following formula:

EXPECTED ANNUAL NET ENERGY PRODUCTION, in kWh

= A x B x C [-D] {SCE Comment: For DERs with Excess-Sales only}

Where:

- A = The Installed DC Rating, in kW_{PDC}. (As of the Effective Date and until SCE's verification of Seller's installation of the Generating Facility pursuant to Exhibit J, this rating is deemed to be [Number] kW_{PDC.}).
- B = [Annual Energy Yield Factor Number] kWh AC per kW_{PDC} per year.
- C = Annual degradation factor ("Annual Degradation Factor") in each Term Year as follows:

Annual Degradation Factor

[D = Expected Annual Site Host Load kWh per Term Year] {SCE Comment: For DERs with Excess-Sales only}

(k) The Generating Facility is a ["small power production facility," as described in 18 CFR §§292.203(a), 292.203(c) and 292.204,] [a "topping-cycle cogeneration facility," as defined in 18 CFR §292.202(d),] [and a "bottoming-cycle cogeneration facility," as defined in 18 CFR §292.202(e)]. {SCE Comment: select all that apply}."

3. Section 1.03(a) is deleted and replaced with the following:

"Subject to any extensions made pursuant to Sections 3.06(f) or 5.03, and further subject to Section 1.03(c), the Commercial Operation Date must be no later than the date that is the first day of the month occurring thirty-six months after CPUC Approval ('Commercial Operation Deadline')."

4. Section 1.03(b) is deleted and replaced with the following:

"Subject to Section 1.03(d), if Seller has not obtained Permit Approval on or before that date that is ninety (90) days before the first day of the month following the date that is thirty-six (36) months from the date of CPUC Approval, then, upon SCE's receipt of Notice from Seller, which Notice must be provided at least sixty (60) days before the date that is thirty-six (36) months from the date of CPUC Approval, the Commercial Operation Deadline shall be extended six (6) months from the first day of the month following the date that is thirty-six (36) months from the date of CPUC Approval; *provided*, *however*, such extension shall not be given if the failure to obtain Permit Approval was as a result of Seller's failure to take all commercially reasonable actions to apply for and meet all of its requirements and deadlines to obtain such Permit Approval."

5. Section 1.05(a) is deleted and replaced with the following:

"Subject to Sections 1.05(b) and 1.05(c)(i), the Product Price is: (i) [Dollar amount text] dollars (\$[Number]]) per MWh, for Subscribed Delivered Energy, (the 'Subscribed Product Price'); and (ii) for each TOD Period, the lesser of (x) the DLAP Price plus the Renewable Energy Credit Market Price, and (y) the Subscribed Product Price multiplied by the applicable Product Payment Allocation Factor, for Unsubscribed Delivered Energy (the 'Unsubscribed Product Price')."

- 6. Section 1.05(b) is amended to add the phrase "or to any Distributed Energy Resource" after the term Generating Facility.
- 7. A new Section 1.05(d) is added as follows:
 - "(d) <u>Deliveries Non-Compliant with the Generation Distribution Factor.</u>

If during any Settlement Interval, Seller delivers Metered Amounts that are inconsistent with the Generation Distribution Factor provided to SCE by Seller, then Seller shall be responsible for and pay any and all CAISO charges, CAISO Sanctions, and CAISO Costs incurred by SCE with respect to such Metered Amounts.".

8. Section 1.08 is deleted in its entirety and replaced with the following: "[Intentionally Omitted.]."

- 9. Section 2.01(b) is deleted and replaced with the following:
 - "(b) Seller's Interconnection Queue Position.

Seller must not (i) withdraw any of the Interconnection Queue Position(s) identified in Section 1.07, (ii) assign or transfer any of the Interconnection Queue Position(s) to any entity, or (iii) utilize any of the Interconnection Queue Position(s) for the benefit of any power purchase and sale agreement other than the Agreement, in each case, without SCE's prior written consent."

- 10. A new Section 2.01(d) is added as follows:
 - "(d) Seller's Community Interest Requirement.

Before the start of any construction activities at the Site, Seller shall demonstrate the Project's compliance with the Community Interest Requirement by providing SCE an attestation in the form of Exhibit R-1. Seller shall provide evidence supporting the attestation in the form of Exhibit R-2 or Exhibit R-3, as applicable. SCE shall evaluate Seller's demonstration and notify Seller whether Seller has satisfied the Community Interest Requirement within thirty (30) days of receiving Seller's attestation and supporting documentation in accordance with this Section 2.01(d).

- 11. Section 2.02(b)(i) is amended to insert the phrase "that is the first calendar day of a month" after the words "the Commercial Operation Date shall be a date".
- 12. Section 2.02(b) is amended to add Section 2.02(b)(7) as follows:
 - "(7) Seller has demonstrated that it has complied with the Residential Requirement. Seller shall provide Buyer with Subscription Information in the format set forth in Exhibit P demonstrating such compliance."
- 13. Section 2.03(b) is deleted and replaced with the following:
 - "(b) Termination Rights of SCE.

SCE has the right to terminate this Agreement on Notice, which will be effective five (5) Business Days after such Notice is given to Seller, on or before the date that is sixty (60) days after Seller provides to SCE the results of any Interconnection Study or interconnection agreement tendered to Seller by the Transmission Provider if:

(i) The Interconnection Studies or agreements associated with the Generating Facility, as of the date of the termination Notice, estimates, includes,

specifies or reflects that the maximum total cost of transmission upgrades or new transmission facilities to SCE, or any Transmission Provider under the jurisdiction of the CAISO, including costs reimbursed by SCE, or any Transmission Provider under the jurisdiction of the CAISO, to Seller ("Aggregate Network Upgrade Costs"), may in the aggregate exceed [dollar amount text] dollars (\$[Number]]) ("Network Upgrades Cap"), irrespective of any subsequent amendments of such Interconnection Studies or agreements or any contingencies or assumptions upon which such Interconnection Studies or agreements are based; or [SCE Comment: Monetary threshold to be based upon transmission-related costs allocated to the Generating Facility that SCE would incur as estimated in the most recent Interconnection Study(ies).}

(ii) SCE must procure transmission service from any other Transmission Provider to allow SCE to Schedule electric energy from the Generating Facility and the cost for such transmission service is not reimbursed or paid by Seller.

Notwithstanding anything to the contrary in this Section 2.03(b), SCE shall have no right to terminate this Agreement under this Section 2.03(b), if Seller, concurrently with its provision of the relevant Interconnection Study or agreement pursuant to Section 3.17(a), irrevocably agrees that Seller shall owe to SCE (I) the amount by which the Aggregate Network Upgrade Costs exceed the Network Upgrades Cap ("Excess Network Upgrade Costs"), and (II) any costs for transmission services specified in Section 2.03(b)(ii). If Seller elects to pay, without reimbursement, for the Excess Network Upgrade Costs pursuant to this Section 2.03(b), in no event shall Seller have any interest in or rights or title to any Network Upgrades (as defined in the CAISO Tariff) or Congestion Revenue Rights (as defined in the CAISO Tariff) in connection with the development of the Generating Facility or the delivery of Product to SCE pursuant to this Agreement."

- 14. Sections 2.04(a)(xiii) and (xiv) are deleted and replaced with the following:
 - "(xiii) Seller's obligations under Sections 1.05(d), 3.01(d)(iv), [3.05(a)(i)(1)] {SCE Comment: for DERs with Shared Facilities only}, 3.32, and 3.35; and
 - (xiv) The obligation of Seller to transfer Green Attributes associated with Product, in accordance with Section 3.01(b), for which SCE has paid the Product Price for or issued a bill credit for.".
- 15. Section 2.04(b) is amended to: (i) add the phrase "lesser of the Subscribed Product Price and the Unsubscribed" to the second paragraph after the phrase "Capacity Attributes and Resource Adequacy Benefits to SCE at the" and before the words "Product Price"; and

- (ii) delete the word "the" and replace it with the word "each" in the first parenthetical of the third paragraph after the word "including" and before the phrase "Interconnection Queue Position".
- 16. Section 3.01(d)(iv) is amended to delete the parenthetical "(which cost shall not be subject to the Compliance Expenditure Cap)".
- 17. Section 3.03 is deleted and replaced with the following: "Throughout the Delivery Term, Seller shall not sell the Product (or any portion thereof) to any entity other than: (i) to SCE, and (ii) to Customers in accordance with the CSA.".
- 18. Section 3.05 is deleted and replaced with the following:
 - "3.05 <u>Permits, Interconnection and Transmission Service Agreements, and CAISO</u> Tariff Compliance.
 - (a) Seller shall obtain and maintain throughout the Delivery Term any and all interconnection and transmission service rights and Permits required to effect delivery of the electric energy from each Distributed Energy Resource to the Distributed Energy Resource Aggregation and from the Distributed Energy Resource Aggregation to the Delivery Point.
 - (b) Seller shall be responsible for all costs and charges directly caused by, associated with, or allocated to the interconnection of each Distributed Energy Resource and the Distributed Energy Resource Aggregation to Transmission Provider's electric system and transmission of electric energy from each Distributed Energy Resource and the Distributed Energy Resource Aggregation to the Transmission Provider's electric system.
 - (c) Seller shall comply with the CAISO Tariff, including securing and maintaining in full force all required CAISO agreements, certifications and approvals.
 - (d) Seller shall secure through the CAISO the CAISO Resource ID that is to be used solely for the Generating Facility.
 - (e) Seller shall comply with the requirements of the CAISO Tariff applicable to Distributed Energy Resources, a Distributed Energy Resource Provider, and a Distributed Energy Resource Aggregation.

{SCE Comment: Language applicable to projects that do not utilize Shared Facilities.}

(a) Seller shall obtain and maintain throughout the Delivery Term any and all interconnection and transmission service rights and Permits required to effect delivery of the electric energy from each Distributed Energy Resource to the

Distributed Energy Resource Aggregation and from the Distributed Energy Resource Aggregation to the Delivery Point. Each interconnection agreement shall provide for interconnection capacity available or allocable to each Distributed Energy Resource that is no less that the applicable DER Contract Capacity. The ownership and use of the Shared Facilities (including the interconnection agreement itself) are or will be subject to a co-tenancy or similar sharing agreement (collectively, "Shared Facilities Agreement(s)"), under which Shared Facilities Agreements an Affiliate of Seller may act as a manager on behalf of Seller and the Other Seller(s) under the interconnection agreement ("Affiliate Manager"). Seller shall ensure that, from the Effective Date and continuing throughout the Term, Seller shall have sufficient interconnection capacity and rights under or through the interconnection agreement and the Shared Facilities Agreements, if any, to interconnect the particular DERs that are sharing facilities with the CAISO-Controlled Grid, and to fulfill its obligations under this Agreement. In connection with each interconnection agreement and each Shared Facilities Agreement, the following shall apply:

(i) The Shared Facilities Agreements shall provide that:

- (1) the Other Seller(s), the Affiliate Manager and the Interconnection Affiliate (if different from the Seller or Other Seller(s)) shall fully indemnify Seller, SCE, and Seller's Customers for any liability arising out of its respective acts or omissions in regards to its respective performance obligations under the interconnection agreement and any Shared Facilities Agreement in which such party is a counterparty with Seller,
- (2) Seller shall have the right to correct, remedy, mitigate, or otherwise cure any omission, failure, breach or default by Other Seller, Affiliate Manager, or Interconnection Affiliate (if different from the Seller or Other Seller(s)) that would negatively impact Seller's obligations under this Agreement, under the interconnection agreement, or under any Shared Facilities Agreement in which Seller is a counterparty, and
- (3) any instruction from the CAISO or Transmission Provider to curtail energy deliveries shall be allocated between the particular DERs that are sharing facilities and the Other Generating Facility(ies) on a pro rata basis based upon installed capacity, except when such pro rata allocation would be in violation of the applicable curtailment instruction.
- (ii) Seller shall, or shall cause the Interconnection Affiliate (if different from Seller), to apply for and expeditiously seek FERC's acceptance of any Shared Facilities Agreement(s), if required.

- (iii) Seller shall not assign or transfer Seller's rights or obligations under the interconnection agreement or any Shared Facilities Agreement to any person or entity without the prior written consent of SCE, which consent shall not be unreasonably withheld.
- (b) As between SCE and Seller under this Agreement, Seller shall be responsible for all costs and charges directly caused by, associated with, or allocated to Seller, the Interconnection Affiliate, the Affiliate Manager, or the Other Seller(s) under the interconnection agreement, the Shared Facilities Agreement, if any, and the CAISO Tariff, in connection with the interconnection of the Generating Facility to the Transmission Provider's electric system and transmission of electric energy from the Generating Facility to the Transmission Provider's electric system.
- (c) Seller shall, or shall cause the Interconnection Affiliate, as applicable, to comply with the CAISO Tariff, including securing and maintaining in full force and effect all required CAISO agreements, certifications and approvals.
- (d) Seller shall, or shall cause the Interconnection Affiliate, as applicable, to secure through the CAISO the CAISO Resource ID that is to be used solely for this Generating Facility.
- (e) Seller shall, or shall cause the Interconnection Affiliate, as applicable, to comply with the requirements of the CAISO Tariff applicable to Distributed Energy Resources, a Distributed Energy Resource Provider, and a Distributed Energy Resource Aggregation, or each applicable equivalent successor.
- (f) Seller shall, or shall cause the Interconnection Affiliate to, as applicable, comply with the metering requirements of the CAISO Tariff applicable to a: (i)
 Distributed Energy Resource Aggregation, (ii) Distributed Energy Resource
 Provider, (iii) Distributed Energy Resource, and (iv) Scheduling Coordinator for a Distributed Energy Resource Aggregation, or each applicable equivalent successor, for the Generating Facility.

{SCE Comment: Language applicable to projects that utilize Shared Facilities.}".

- 19. Section 3.06(a) (c) are deleted and replaced with the following:
 - "3.06 Modification of Special Conditions.
 - (a) If the Contract Capacity or a DER Contract Capacity, each as set forth in Section 1.01(h) is greater than the Demonstrated Contract Capacity or the applicable DER Demonstrated Contract Capacity, respectively, then:
 - (i) The Contract Capacity or DER Contract Capacity, as applicable, will be reduced to an amount equal to the Demonstrated Contract Capacity or the applicable DER Demonstrated Contract Capacity;

- (ii) The Expected Annual Net Energy Production will be recalculated using such adjusted Contract Capacity; and
- (iii) The Performance Assurance Amount for the Performance Assurance required to be posted and maintained pursuant to Section 8.03 will be recalculated using such adjusted Contract Capacity, and any amount of Performance Assurance in excess of that required for the adjusted Contract Capacity will be returned to Seller.

 [SCE Comment: For all technologies except Solar Photovoltaic]
- (b) If the Installed DC Rating or the DER Installed DC Rating, each as set forth in Section 1.01(i) is greater than the Demonstrated Installed DC Rating or the DER Demonstrated Installed DC Rating, respectively,
 - (i) The Installed DC Rating or the DER Installed DC Rating, as applicable, will be reduced to an amount equal to the Demonstrated Installed DC Rating or the DER Demonstrated Installed DC Rating, as applicable;
 - (ii) The Expected Annual Net Energy Production will be recalculated using such adjusted Installed DC Rating; and
 - (iii) The Performance Assurance Amount for the Performance Assurance required to be posted and maintained pursuant to Section 8.03 will be recalculated using such adjusted Installed DC Rating, and any amount of Performance Assurance in excess of that required for the adjusted Installed DC Rating will be returned to Seller.

{SCE Comment: For Solar Photovoltaic}

- (c) Neither Party will have any liability for failure to purchase or deliver Product associated with or attributable to capacity in excess of the Demonstrated Installed DC Rating, the DER Demonstrated Installed DC Rating, the Demonstrated Contract Capacity, or the DER Demonstrated Contract Capacity ("Unincluded Capacity"), subject to Section 3.06(d).".
- 20. Section 3.06(g) is deleted and replaced with the following:
 - "Subject to Section 8.02(c)(i) and the Commercial Operation Date occurring on or before the Commercial Operation Deadline or any extended Commercial Operation Deadline as provided in this Agreement, if Seller demonstrates the full [Installed DC Rating specified in Section 1.01(i) and the full DER Installed DC Rating for each Distributed Energy Resource specified in Section 1.01(i)] {SCE Comment: For Solar Photovoltaic} [Contract Capacity specified in Section 1.01(h) and the DER Contract Capacity for each Distributed Energy Resource specified in Section 1.01(h)] {SCE Comment: For all

technologies except Solar Photovoltaic}, in accordance with the procedure set forth in Exhibit J; then SCE shall return the full Development Security.".

- 21. Section 3.07(a)(ii) is amended to delete the parenthetical "(calculated in accordance with Exhibit K)".
- 22. Section 3.08 is deleted and replaced with the following:
 - "3.08 Metering, Communications, Telemetry and Meteorological Station(s).
 - (a) <u>CAISO Approved Meter and CAISO Approved DER Meter.</u>

Seller shall, at its own cost, install, maintain and test all CAISO Approved DER Meter(s) for each Distributed Energy Resource and CAISO Approved Meters for the Distributed Energy Resource Aggregation.

{SCE Comment: Language applicable to projects that do not use a shared transformer.}

(b) <u>Check Meter</u>.

Seller will permit SCE to furnish and install one Check Meter on the high voltage side of the step-up transformer, substation, or any other location at SCE's sole discretion, associated with each Distributed Energy Resource and the Distributed Energy Resource Aggregation in compliance with the applicable utility electric service requirements. Each Check Meter must be interconnected with SCE's communication network to permit:

- (i) Periodic, remote collection of revenue quality meter data; and
- (ii) Back-up real time transmission of operating-quality meter data.

{SCE Comment: Language applicable to projects that do not use a shared transformer.}

(c) SCE's Access to Meters.

- (i) Subject to Section 3.18, Seller hereby grants SCE reasonable access to all CAISO Approved DER Meter(s), CAISO Approved Meters and Check Meters for meter readings and any purpose necessary to effectuate this Agreement.
- (ii) Seller shall promptly provide SCE access to all meter data and data acquisition services both in real-time, and at later times, as SCE may reasonably request.
- (iii) Prior to the Commercial Operation Date, Seller shall provide instructions to the CAISO granting authorizations or other documentation sufficient to provide SCE with access to the CAISO Approved Meter(s), CAISO Approved DER Meter(s), and to Seller's settlement data on OMAR.

(d) <u>CAISO Approved Meter Maintenance and CAISO Approved DER Meter Maintenance.</u>

- (i) Seller shall test and calibrate the CAISO Approved DER Meter(s) and the CAISO Approved Meter(s), as necessary, but in no event will the period between testing and calibration dates be greater than twelve (12) months.
- (ii) Seller shall bear its own costs for any meter check or recertification of the CAISO Approved DER Meter(s) and the CAISO Approved Meter(s).
- (iii) Seller shall replace each CAISO Approved DER Meter(s) and the CAISO Approved Meter battery at least once every thirty-six (36) months or such shorter period as may be recommended by the CAISO Approved DER Meter(s) or CAISO Approved Meter manufacturer, as applicable.
 - Notwithstanding the foregoing, if a CAISO Approved DER Meter or a CAISO Approved Meter battery fails, Seller shall replace such battery within one (1) day after becoming aware of its failure.
- (iv) Seller shall use certified test and calibration technicians to perform any work associated with the CAISO Approved DER Meter(s) and the CAISO Approved Meter(s).
- (v) Seller shall inform SCE of test and calibration dates, provide SCE with access to observe and witness such testing and calibration, and provide SCE certified results of tests and calibrations within thirty (30) days after completion.

(e) SCADA and Telemetry System.

All communication, metering, telemetry, and associated generation operation equipment will be centralized into the [Distributed Energy Resource Aggregation's] {SCE Comment: For DERs} [Generating Facility's] {SCE: Comment: For non-DER Generating Facilities} SCADA. Seller shall configure the SCADA so that SCE may access it via the GMS from the GOC. Seller shall link the systems via an approved SCE communication network, utilizing existing industry standard network protocol, as approved by SCE. The connection will be bidirectional in nature and used by the Parties to exchange all data points to and from the GOC. Seller shall be responsible for the costs of installing, configuring, maintaining and operating the SCADA and internal site links for the Distributed Energy Resource Aggregation.

Seller shall be responsible for designing, furnishing, installing, operating, maintaining and testing a real time Telemetry System capable of interconnecting to the GMS, the CAISO Approved Meter(s) and the Distributed Energy Resource Aggregation's control system with the CAISO's Energy Communication Network.

Seller shall be responsible for designing, furnishing, installing, operating, maintaining and testing a real time DER Telemetry System capable of allowing the Distributed Energy Resource Aggregation's control system to interconnect to and control each Distributed Energy Resource, permitting each Distributed Energy Resource to comply with any CAISO dispatch or curtailment pursuant to Section 3.12(g).

The Telemetry System shall be designed in accordance with the CAISO monitoring and communication requirements and must be capable of:

- (i) Reporting data from each CAISO Approved Meter;
- (ii) Providing the status of key control points from the Distributed Energy Resource Aggregation's control system;
- (iii) Routing generating unit set points to the Distributed Energy Resource Aggregation's control system; and
- (iv) Communicating availability of the Generating Facility pursuant to Section 3.08(g).

The Telemetry System must include a data processing gateway, internet connection, interconnecting cabling and all service agreements required for accessing the CAISO's Energy Communications Network.

The above mentioned connections and data transfer must be included in the systems engineering tasks as a part of the construction of the Generating Facility, and must be fully functional before Commercial Operation.

The DER Telemetry System shall be designed in accordance with the CAISO monitoring and communication requirements and must be capable of complying with the requirements set forth in CAISO Tariff applicable to a Distributed Energy Resource, a Distributed Energy Resource Provider, a Distributed Energy Resource Aggregation, and any requirement applicable to a Scheduling Coordinator for a Distributed Energy Resource Aggregation.

(f) Meteorological Station(s) and Reporting Requirements.

If required by the CAISO, Seller, at its own expense, shall install and maintain one (1) or more stand-alone meteorological stations at each Distributed Energy Resource in accordance with Exhibit O to monitor and report weather data to both the CAISO and the existing SCE weather station data collection system.

The station(s) must be installed at least sixty (60) days before Commercial Operation.

The station(s) must be equipped with the Meteorological Equipment, as may be modified by Seller at SCE's direction from time to time to reflect the CAISO's PIRP/EIRP protocol and the requirements of Exhibit K.

The station(s) must be designed to collect and record data in accordance with CAISO's PIRP/EIRP protocols and the requirements of Exhibit K.

Data reports must be formatted in a manner consistent with the CAISO requirements published on the CAISO internet website.

Telemetering equipment must be designed to function in accordance with CAISO's PIRP/EIRP protocols.

The station(s) must be equipped to measure and record the minimum data required by the CAISO, in the manner specified by the CAISO.

Seller shall submit to SCE for review and approval, Seller's technical specifications for the meteorological station(s) along with a site plan showing the location of the station(s), the location of all [Wind Turbines, the wind rose for the Site], [Solar Generating Units, Photovoltaic Modules, Current Inverters,] and other prominent features, as applicable.

{SCE Comment: Intermittent only.}

Seller shall calibrate all first and second class thermopile pyranometers to the same nationally recognized standard and apply temperature correction to the measurement. Seller's Telemetry System shall transmit the calibrated data to SCE. Such temperature correction shall be based upon a calibration of the actual instrument or to a generic temperature curve that is supported by data from a nationally recognized testing laboratory to be representative of the exact vintage and model of instruments to be used by the Seller. Furthermore, Seller shall calibrate all thermopile pyranometers, regardless of type, every Term Year and upon SCE's reasonable request.

{SCE Comment: Solar photovoltaic only.}

- (g) Real-Time Communication of Availability.
 - (i) Prior to the Commercial Operation Date, Seller shall install a telecommunication system and demonstrate to SCE's reasonable satisfaction that the system interfaces with the Web Client and the GMS to provide SCE with Seller's Real-Time Availability.
 - (ii) Seller shall maintain the telecommunications path, the hardware, and software to provide quality data to SCE throughout the Delivery Term.
 - (iii) Upon Notice from SCE, Seller shall repair or have corrected as soon as possible, but no later than five (5) days after receipt of such Notice any:
 - (1) Inoperable telecommunications path;
 - (2) Inoperable software; or
 - (3) Faulty instrumentation.
 - (iv) Seller shall submit an Actual Availability Report pursuant to Section 3.22

for any month in which Seller's telecommunications system was not available or did not provide quality data for longer than twenty-four (24) continuous hours."

- 23. Section 3.09(a) is deleted and replaced with the following:
 - "(a) This Agreement is DER Site specific as set forth in Section 1.01(b). Seller may change the location of any DER Site only upon SCE's prior written consent, which consent is in SCE's sole discretion."
- 24. Section 3.11(a) is deleted and replaced with the following:
 - "(a) Designing and constructing the Generating Facility and the Distributed Energy Resource Aggregation;".
- 25. Section 3.11(c)(i) is deleted and replaced with the following:
 - "(i) Site plan drawings for the Generating Facility and the Distributed Energy Resource Aggregation;".
- 26. Section 3.11(c)(vi) is deleted and replaced with the following:
 - "(vi) Longitude and latitude of the centroid of each DER Site;".
- 27. Section 3.12 is deleted and replaced with the following:
 - "3.12 Operation and Record Keeping.
 - (a) Seller shall maintain all Permits, licenses, certifications, and approvals necessary for the operation and maintenance of the Generating Facility and the Distributed Energy Resource Aggregation in accordance with Prudent Electrical Practices, Applicable Laws, Permit Requirements, and Industry Standards.
 - (b) Seller shall comply with Operating orders in compliance with the CAISO Tariff and any interconnection agreement.
 - (c) On or prior to the Commercial Operation Date:
 - (i) SCE shall have obtained or waived CPUC Approval;
 - (ii) Seller shall obtain CEC Pre-Certification;
 - (iii) Seller shall take all steps necessary to ensure that SCE becomes authorized by the CAISO to Schedule the electric energy produced by the Generating Facility and aggregated by Seller to the Distributed Energy Resource Aggregation with the CAISO;

- (iv) SCE shall have been authorized by the CAISO to Schedule the electric energy produced by the Generating Facility and aggregated by Seller to the Distributed Energy Resource Aggregation with the CAISO;
- (v) Seller shall demonstrate to SCE's reasonable satisfaction that Seller has executed all necessary Transmission Provider and CAISO agreements, including those applicable to a Distributed Energy Resource, a Distributed Energy Resource Provider, and a Distributed Energy Resource Aggregation;
- (vi) Seller shall provide to SCE each DLF used by the Transmission Provider in the administration of the transmission service agreement for the Generating Facility;
- (vii) Seller shall be Forecasting to SCE in accordance with Exhibit D;
- (viii) Seller shall commence delivering electric energy to SCE at the Delivery Point and the Generating Facility is operating in parallel with Seller's Transmission Provider;
- (ix) Seller shall have installed and placed in operation all equipment and systems required under Section 3.08; {SCE Comment: Intermittent only.}
- (x) Seller shall have registered with the NERC as the Generating Facility's Generator Owner and Generator Operator if Seller is required to be a registered entity pursuant to the NERC Reliability Standards;
- (xi) Seller shall have furnished to SCE all insurance documents required under Section 10.11;
- (xii) Seller shall have delivered to Buyer no later than sixty (60) days prior to the anticipated Commercial Operation Date the Subscription Information and Bill Credit Instructions required under Section 3.30 containing the information required by such report;
- (xiii) Buyer shall have confirmed in writing that it has verified, with respect to each Subscribed Customer listed in the Subscription Information and Bill Credit Instructions delivered pursuant to Section 3.12(c)(xii) that: (x) such Customer has enrolled in Buyer's CR Tariff; and (y) the Subscription amount for such Customer (1) does not exceed one hundred twenty percent (120%) of such Customer's forecasted annual load, as such load is reasonably determined by Buyer based on historical usage data, and (2) is projected to be an amount of energy per year equal to or greater than: (A) 100 kWh per month on average, calculated on an annual basis or (B) twenty five percent (25%) of such Customer's load, or as otherwise required by Green-e® Energy;

- (xiv) Seller shall deliver to Buyer an original legal opinion, in form and substance acceptable to Buyer, and addressed to Buyer. The legal opinion shall state that the transactions between the Customers and Seller either (x) do not involve the offer or sale of 'securities' under California or federal law, or, (y) to the extent that such transactions involve the offer or sale of securities under California or federal law, the transactions (1) involve the offer or sale of securities that are registered under federal securities law and exempt from qualification under California securities law, (2) involve the offer or sale of securities that are registered under federal securities law and are qualified under California securities law, (3) involve the offer or sale of securities that are exempt from registration under federal securities law and are qualified under California securities law, or (4) involve the offer or sale of securities that are exempt from registration under federal securities law and exempt from qualification under California securities law, as applicable. The legal opinion may not contain any exceptions or qualifications unacceptable to Buyer in its reasonable discretion. The Seller must submit to Buyer an attestation from an officer of Seller that the fact certificate provided by an officer of the Seller to the law firm issuing the legal opinion is true and complete and that Seller's business model with Customers is, and throughout the Delivery Term will be, as described in the legal opinion;
- (xv) Seller shall take all steps necessary to ensure that SCE in its capacity as Scheduling Coordinator shall have access to the CAISO Approved DER Meter data;
- (xvi) Seller shall provide SCE with a copy of the Master File for the Distributed Energy Resource Aggregation;
- (xvii) Seller shall provide SCE with a copy of Seller's Generation Distribution Factor, and
- (xviii) Seller shall demonstrate to SCE's reasonable satisfaction that the Generating Facility and each DER is a Qualifying Facility.
- (d) Seller shall keep a daily operations log for the Generating Facility that shall include the following information:
 - (i) Availability of the Generating Facility; {SCE Comment: All technologies except Solar Photovoltaic.}

Availability of the Inverter Block Units and associated Current Inverters; {SCE Comment: Solar Photovoltaic only.}

- (ii) Circuit breaker trip operations;
- (iii) Any significant events related to the Operation of the Generating Facility;

- (iv) Real and reactive power and energy production;
- (v) Changes in Operating status;
- (vi) Protective apparatus operations;
- (vii) Any unusual conditions found during inspections;
- (viii) Electric energy production, fuel consumption and efficiency (if applicable); and
- (ix) Status and settings of generator controls including automatic voltage regulator and power system stabilizer.
- (e) Seller shall log changes in the generator output setting if it is "block-loaded" to a specific kW capacity.

{SCE Comment: Baseload only.}

Seller shall maintain complete records of the Generating Facility's wind speeds and other pertinent meteorological conditions and operational status of each Wind Turbine.

{SCE Comment: Wind only.}

Seller shall maintain complete records of the Generating Facility's direct normal insolation, other pertinent meteorological conditions and operational status of each Solar Distributed Energy Resource.

{SCE Comment: Solar Thermal only.}

Seller shall maintain complete records of the Generating Facility's plane of array insolation, other pertinent meteorological conditions and operational status of each Inverter Block Unit.

{SCE Comment: Solar Photovoltaic only.}

Seller shall maintain complete records of the Generating Facility's fuel consumption if a biomass or landfill generating facility, or geothermal fluid consumption if a geothermal generating facility.

{SCE Comment: Biomass and Geothermal only.}

(f) Seller shall keep a maintenance log for the Generating Facility that shall include information on maintenance (both breakdown and preventative) performed, outages, inspections, manufacturer recommended services and replacement, electrical characteristics of the generators, control settings or adjustments of equipment and protective devices.

Seller shall maintain documentation of all procedures applicable to the testing and maintenance of the Generating Facility protective devices as necessary to comply with NERC Reliability Standards applicable to protection systems for large electric generators, if Seller is required to be a registered entity pursuant to the NERC Reliability Standards.

- (g) Seller shall promptly curtail the production of the Generating Facility:
 - (i) Upon Notice from SCE that Seller has been instructed by the CAISO or Transmission Provider to curtail energy deliveries; *provided*, solely the action of the CAISO issuing a Schedule shall not by itself, without other factors, constitute an instruction by the CAISO to curtail energy deliveries pursuant to this Section 3.12(g)(i);
 - (ii) Upon Notice that Seller has been given a curtailment instruction or similar instruction in order to respond to an Emergency; or
 - (iii) If SCE, in its capacity as Buyer under this Agreement, issues a Curtailment Order.
- (h) Information maintained pursuant to this Section 3.12 shall be kept by Seller throughout the Delivery Term and for four (4) years thereafter, and shall be provided or made available to SCE within twenty (20) days after SCE's request.
- (i) Seller must be interfaced with SCE's Generation Management System and be responsive to 5-minute Dispatch Instruction and other applicable CAISO Tariff rules.".
- 28. Section 3.13 is deleted and replaced with the following:

"3.13 Obtaining Scheduling Coordinator Services.

Seller shall comply with all applicable CAISO Tariff procedures, protocol, rules and testing as necessary for SCE to submit Bids for the electric energy produced by the Generating Facility and the Distributed Energy Resource Aggregation.

- (a) Designating SCE as Scheduling Coordinator.
 - (i) At least thirty (30) days before the Commercial Operation Date, Seller shall take all actions and execute and deliver to SCE all documents necessary to authorize or designate SCE, or SCE's designee, as the Scheduling Coordinator throughout the Delivery Term.
 - (ii) Throughout the Delivery Term, Seller shall not authorize or designate any other party to act as Scheduling Coordinator, nor shall Seller perform, for its own benefit, the duties of Scheduling Coordinator.
 - (iii) Seller is responsible for and shall pay SCE an amount equal to the costs (including the costs of SCE employees or agents) SCE incurs, as determined in SCE's sole discretion, as a result of SCE being designated as the Generating Facility's Scheduling Coordinator including the costs associated with the registration of the Distributed Energy Resource Aggregation with the CAISO, and the installation, configuration, and testing of all equipment and software necessary for SCE to act as

Scheduling Coordinator or to Schedule the Generating Facility ("SC Set-up Fee"); *provided*, the SC Set-up Fee shall not exceed \$20,000.

(b) Replacement of SCE as Scheduling Coordinator.

At least forty-five (45) days before the end of the Term, or as soon as practicable before the date of any termination of this Agreement before the end of the Term, Seller shall take all actions necessary to terminate the designation of SCE, or SCE's designee, as Seller's Scheduling Coordinator as of hour ending 24:00 on the last day of the Term. These actions include the following:

- (i) Seller shall submit to the CAISO a designation of a new Scheduling Coordinator for Seller to replace SCE;
- (ii) Seller shall cause the newly designated Scheduling Coordinator to submit a letter to the CAISO accepting the designation; and
- (iii) Seller shall inform SCE of the last date on which SCE will be Seller's Scheduling Coordinator. SCE must consent to any date other than the last day of the Term, such consent not to be unreasonably withheld.".
- 29. Section 3.17 is deleted and replaced with the following:

"3.17 Provision of Information.

Seller shall promptly provide to SCE copies of:

- (a) Within ten (10) Business Days of receipt thereof, copies of any Interconnection Study or the interconnection agreement tendered to Seller by the Transmission Provider and, concurrently with the provision of the first Interconnection Study or interconnection agreement tendered to Seller by the Transmission Provider that may give rise to a termination right of SCE under Section 2.03(b), Seller shall also provide SCE a Notice of Seller's irrevocable election to exercise or not exercise its right to assume financial responsibility for any Excess Network Upgrade Costs pursuant to Section 2.03(b), with a failure to provide such an election deemed to be an election not to exercise such rights;
- (b) All applications and approvals or disapprovals relating to CEC PreCertification, CEC Certification, CEC Verification, any Permit and PIRP/EIRP (in the event SCE requests Seller to apply to be in PIRP/EIRP);
- (c) All final and revised copies of material reports, studies and analyses furnished by the CAISO or any Transmission Provider, and any correspondence related thereto, concerning the interconnection of the Generating Facility to the Transmission Provider's electric system or the transmission of electric energy on the Transmission Provider's electric system;
- (d) All notifications of adjustments in the DLF used by the Transmission Provider in the administration of the transmission service agreement for the Generating

Facility within thirty (30) days of receiving such notification from the Transmission Provider;

(e) A copy of the Final Wind Report, and any updates thereafter for the time period beginning on the Effective Date and ending on the last day of the first Term Year; {SCE Comment: Wind only.}

All Geothermal Reservoir Reports, and any revisions thereto, for the time period beginning on the Effective Date and ending on the last day of the first Term Year; {SCE Comment: Geothermal only.}

All Solar Resource Evaluation Reports, and any revisions thereto, for the time period beginning on the Effective Date and ending on the last day of the first Term Year;

{SCE Comment: Solar only.}

- (f) Any reports, studies, or assessments of the Generating Facility prepared for Seller by an independent engineer; and
- (g) All Generating Facility, Distributed Energy Resource Aggregation, and metering information as may be requested by SCE, including the following, at least thirty (30) days before the Commercial Operation Date:

For each CAISO Approved DER Meter and CAISO Approved Meter:

- (i) Generating Station/Unit ID;
- (ii) CAISO Resource ID;
- (iii) CAISO Approved Meter Device ID;
- (iv) Password;
- (v) CAISO Approved DER Meter Device ID;
- (vi) Data path (network (ECN) or modem);
- (vii) If modem, phone number;
- (viii) Copy of meter certification(s);
- (ix) List of any CAISO metering exemptions (if any); and
- (x) Description of any compensation calculations such as transformer losses and line losses.

For the Generating Facility:

- (1) Utility transmission/distribution one line diagram;
- (2) Physical location, address or descriptive identification;
- (3) Latitude and longitude of [the centroid and each corner of each DER] {SCE Comment: For solar only} [each Distributed Energy Resource]

{SCE Comment: For all other technologies} [, and all Meteorological Equipment]; {SCE Comment: Intermittent only.}

- (4) Telephone number on site;
- (5) Telephone number of control room;
- (6) Telephone number for operational issues; and
- (7) Telephone number for administrative issues.
- (h) The names of the Interconnection Point(s) and the Delivery Point within thirty (30) days after Seller's receipt of such information from the Transmission Provider or CAISO, as applicable.

{SCE Comment: Applicable if the official names of the Interconnection Point or Delivery Point are not known as of the Effective Date.}

- (i) No later than twenty (20) days after each semi-annual period ending on June 30th or December 31st, a report listing all Diverse Business Enterprises that supplied goods or services to Seller during such period, including any certifications or other documentation of such Diverse Business Enterprises status as such and the aggregate amount paid to Diverse Business Enterprises during such period.
 - (i) SCE has the right to disclose to the CPUC all such information provided by Seller pursuant to this Section 3.17(i).
 - (ii) Seller shall make reasonable efforts to accommodate requests by the CPUC (or by SCE in response to a request by the CPUC) to audit Seller in order to verify data provided by Seller pursuant to this Section 3.17(i).
- (j) Internal Revenue Service tax Form W-9 and California tax Form 590 (or their equivalent), completed with Seller's information, and any other documentation necessary for SCE to comply with its tax reporting or withholding obligations with respect to Seller, within ten (10) Business Days of Seller's receipt of Notice from SCE requesting the same.
- (k) [Intentionally Omitted].
- (l) Settlement Quality Meter Data for each Distributed Energy Resource and for the Distributed Energy Resource Aggregation, within ten (10) Business Days of Seller's receipt of Notice from SCE requesting the same.
- (m) Seller shall keep all operating records required of a Qualifying Facility by any applicable CPUC order as well as any additional information that may be required of a Qualifying Facility in order to demonstrate compliance with all applicable standards which have been adopted by the CPUC.

- (n) If the Generating Facility is a 'qualifying cogeneration facility' as contemplated in 18 CFR Section 292.205, then within thirty (30) days following the end of each calendar year, and within thirty (30) days following the end of the Delivery Term, Seller shall provide to Buyer:
 - (i) A copy of a FERC order waiving for the Generating Facility, the applicable operating and efficiency standards for qualifying cogeneration facilities for the applicable year; or
 - (ii) A completed copy of Buyer's 'QF Efficiency Monitoring Program Cogeneration Data Reporting Form,' substantially in the form of Exhibit Q-1, with calculations and verifiable supporting data, which demonstrates the compliance of the Generating Facility with cogeneration Qualifying Facility operating and efficiency standards set forth in 18 CFR Section 292.205 'Criteria for Qualifying Cogeneration Facilities,' for the applicable year.
- (o) If the Generating Facility is a 'qualifying small power production facility' as contemplated in 18 CFR Section Sections 292.203(a), 292.203(c) and 292.204, then within thirty (30) days following the end of each year, and within thirty (30) days following the end of the Delivery Term, Seller shall provide to Buyer:
 - (i) A copy of a FERC order waiving for the Generating Facility, the applicable operating and fuel use standards for qualifying small power production facilities for the applicable year; or
 - (ii) A completed copy of Buyer's 'Fuel Use Standards Small Power Producer Data Reporting Form,' substantially in the form of Exhibit Q-2, with calculations and verifiable supporting data, which demonstrates the compliance of the Generating Facility with small power producer Qualifying Facility fuel use standards set forth in 18 CFR Section 292.204 'Criteria for Qualifying Small Power Production Facilities,' for the applicable year."
- 30. Section 3.19(a) is deleted and replaced with the following:
 - "(a) Within thirty (30) days after the Commercial Operation Date, Seller shall file an application or other appropriate request with the CEC for CEC Certification for each Distributed Energy Resource, the Distributed Energy Resource Aggregation, or the Generating Facility, as applicable."
- 31. Section 3.21(c) is deleted and replaced with the following:
 - "(c) Product Replacement Damage Amount Calculation.

The Lost Output amount that will be used by SCE in the Product Replacement Damage Amount calculation, set forth in Exhibit F, will be the amount reasonable determine by SCE.".

- 32. The first paragraph of Section 3.22(a) is deleted and replaced with the following:
 - "(a) Throughout the Delivery Term, Seller shall prepare and provide to SCE a report with the Actual Available Capacity of the Generating Facility and for each Distributed Energy Resource (an "Actual Availability Report") for each month.".
- 33. [Intentionally Omitted.] {SCE Comment: for all technologies other than Wind} [Section 3.23(a) is deleted and replaced with the following:
 - "(a) Seller shall provide to SCE a minimum of one (1) year of recorded meteorological data from each DER Site not later than ninety (90) days before the Commercial Operation Date.
 - Seller may provide data from additional years if any such data is available.".] {SCE Comment: For Wind only}
- 34. [Intentionally Omitted.] {SCE Comment: for all technologies other than Solar} [The first paragraph of Section 3.24(a) is deleted and replaced with the following:
 - "(a) Seller shall provide to SCE a minimum of one (1) year of recorded meteorological data from each DER Site not later than ninety (90) days before the Commercial Operation Date.".] {SCE Comment: For Solar only}
- 35. New Sections 3.30, 3.31, 3.32, 3.33, 3.34, 3.35, 3.36, 3.37 and 3.38 are added as follows:

"3.30 Subscription.

Seller shall provide Buyer with Subscription Information and Bill Credit Instructions in the format set forth in Exhibit P (as such Exhibit P may be modified by the Buyer in its reasonable discretion to reflect updates to its business practices) setting forth, with respect to each of Seller's Customers for the Project, the information required in Exhibit P, such Subscription Information and Bill Credit Instructions to be delivered no later than sixty (60) days prior to the Commercial Operation Date and, thereafter, ten (10) Business Days after the first day of each calendar month, with respect to the prior calendar month. Following the Effective Date, Seller may include only Customers for the Project who are located in Buyer's then-current service territory on Seller's Subscription Information and Bill Credit Instructions. If Seller becomes aware of incorrect information contained in any current or previously submitted Subscription Information and Bill Credit Instructions, Seller shall provide Buyer with updated Subscription Information and Bill Credit Instructions. Buyer shall not be liable

for any action it takes or fails to take based on incorrect information contained in inaccurate or incomplete Subscription Information and Bill Credit Instructions.

3.31 Assignment of Payments.

Throughout the Delivery Term, Seller shall assign the right to payments for Subscribed Delivered Energy to its Customers.

3.32 Green-e® Energy.

- (a) Throughout the Delivery Term, Seller shall:
 - (i) Comply with the Green-e® Energy eligibility criteria, requirements and best practices as updated from time to time by Green-e® Energy;
 - (ii) Provide all forms, disclosures and other documentation required by Buyer and its auditors in connection with the annual Green-e® Energy verification and audit;
 - (iii) Provide to Buyer a copy of all annual Disclosure Documents that it provides to Customers;
 - (iv) Provide Buyer with a completed 'Green-e® Energy Attestation From Generator Participating In A Tracking System' form (or successor form available on Green-e® Energy's website) promptly when required by Buyer; and
 - (v) Provide Buyer with Green-e® Energy host attestations as they are requested.
- (b) Throughout the Delivery Term and surviving expiration of the Agreement, Seller shall disclose information requested by Buyer or Green-e® Energy for Green-e® Energy certification, including but not limited to information related to:
 - (i) Seller agreeing to provide Green-e® Energy certified resources to all Customers;
 - (ii) Seller agreeing to abide by Green-e® Energy requirements and best practices as specified on the Green-e® Energy website;
 - (iii) Ensuring that all marketing of and disclosures relating to the Project is accurate and in compliance with the FTC Green Guides, and Green-e® Energy requirements, the Green-e® Energy Code of Conduct, and best practices;
 - (iv) Seller maintaining a webpage with disclosures about the Project, Seller's customer service contact information, and links to both

- Buyer's CR webpage and the Green-e® Energy website;
- (v) Providing completed Disclosure Documents to each potential Customer prior to signing a CSA with a customer and in a welcome packet distributed sixty (60) days prior to the Commercial Operation Date and annually thereafter (and in each case with a copy to Buyer), along with a statement that such Disclosure Documents are required by Green-e® Energy, which shall include, without limitation: (1) the amount of energy, in kWh, that Customer has been provided from the Project; (2) the price per kW or kWh; (3) the kW or kWh contracted for (option to also include percentage of Generating Facility's output); (4) the Term; (5) the renewable resource mix; (6) the Generating Facility location; (7) Seller's contact information; (8) a disclaimer stating that capacity does not guarantee a certain amount of output and output may vary (if selling in kW); (9) an estimated output in kWh for each Customer's Subscription (if selling in kW); (10) the average kW needed to power a home in the region (if selling in kW); (11) Seller's customer service contact information; (12) a link to Buyer's CR webpage; (13) all terms and conditions of Customer's Subscription; and (14) a statement that these disclosures are required by Green-e® Energy and information about Green-e® Energy certification and link to Green-e® Energy's website: www.green-e.org/energy; and
- (vi) Seller's compliance with Green-e® Energy's annual verification and audit.

3.33 Marketing Requirements.

Prior to the Effective Date and continuing throughout the Delivery Term: (a) Seller shall have complied with and shall continue to comply with SCE's marketing requirements, as set forth on SCE's website, at www.on.sce.com/CRDevInfo, or successor thereof, and as may be modified by SCE from time to time, and any Green-e® Energy eligibility criteria and requirements; (b) all marketing by Seller shall be accurate and in compliance with the FTC Green Guides; (c) any changes to the marketing plan or marketing materials shall be submitted to Buyer for review prior to Seller's use of such materials; and (d) Seller shall maintain an internet website dedicated to the Project containing disclosures about the Project as required by Green-e® Energy, including a link to Buyer's CR Tariff webpage, a link to the Green-e® Energy website, and customer service contact information.

3.34 California Air Resources Board's Voluntary Renewable Electricity Program.

Throughout the Delivery Term, the Seller shall ensure that the Project complies with the requirements of the California Air Resources Board's Voluntary Renewable Electricity Program and Seller shall provide Buyer with all documents necessary to enable Buyer to retire greenhouse gas allowances on behalf of Customers in compliance with the California Air Resources Board's Voluntary Renewable Electricity Program.

3.35 <u>Customer-Seller Agreement.</u>

Seller shall have a CSA with each Customer, which shall comply with the Greene® Energy eligibility criteria and requirements, and Seller shall be required to include the following provisions in each CSA:

- (a) An outline detailing the program structure of the CR Tariff, including the bill credit mechanism and a statement that Buyer is not a party to, or third party beneficiary of, the CSA or the transactions between Seller and Customer, other than as a conduit for bill credits pursuant to Seller's Subscription Information and Bill Credit Instructions;
- (b) The benefits and risks to Customer of subscribing to the Project, including any termination of the Agreement or termination fees that may be assessed by Seller or Buyer, and that Customer should not expect to receive bill credits in excess of the amount of consideration it provides to Seller under the CSA;
- (c) Customer acknowledgment of the risks associated with participating in wholesale energy markets;
- (d) Customer acknowledgment that it should not have any expectation of profits in deciding to enter into the CSA;
- (e) Customer acknowledgment that it will only receive bill credits to the extent the Project actually generates Qualified Amounts and Seller provides the correct Subscription Information and Bill Credit Instructions to Buyer as specified in Section 3.30;
- (f) The CSA will automatically terminate upon termination or expiration of this Agreement;
- (g) Customer acknowledgment that Buyer is not an issuer or underwriter under California or federal securities laws with respect to the Project, and that Buyer is not making an offer to sell or selling any securities whatsoever;
- (h) All disputes (including those related to bill credits) will be handled between the Seller and Customer pursuant to the dispute resolution provisions in the CSA;

- (i) Customers must enroll with Buyer's CR Tariff as a condition to being eligible to receive bill credits;
- (j) Customers must un-enroll from Buyer's CR Tariff if Customer no longer wishes to subscribe to the Project;
- (k) Customers cannot transfer their Subscriptions to other parties;
- (l) Customers may not subscribe for more than 120% of their forecasted annual load, as reasonably determined by Buyer based on historical usage data;
- (m) Customer Subscription payments to Seller, if any, are refundable until the Commercial Operation Date has been achieved, and Customer subscriptions are portable within Buyer's territory upon the Effective Date;
- (n) Seller shall notify Customer in the event of Seller's imminent bankruptcy or insolvency, or if foreclosure proceedings are initiated on the Project;
- (o) Disclosure that the Customer Subscription may be considered a 'security' issued by Seller under federal or state law;
- (p) Customer is not guaranteed any energy production from the Project;
- (q) Information describing Green-e® Energy and what requirements Seller is subject to in order to provide Customers with Green-e® Energy product;
- (r) A description of Customer access rights to the Site and the Generating Facility, if any;
- (s) Seller and Buyer shall share Customer information amongst themselves for purposes of billing and credits, program eligibility and verifying participation and that Buyer and Seller shall maintain the confidentiality of Customer information:
- (t) Seller's customer service department must respond to Customer inquiries within two (2) Business Days after a Customer request;
- (u) Seller shall indemnify Customers for claims arising from or related to Seller's construction, operation or financing of the Project, including liens of any type, mortgages, stop notices, and claims for bodily injury, death or property damage or destruction;
- (v) Seller will provide Buyer with Subscription Information and Bill Credit Instructions related to the Subscribed capacity, and Seller shall indemnify Buyer for all related claims and billing disputes between Customer and Seller. All bill credits to Customer shall be subject to set-off and counterclaim by Buyer under Seller's power purchase agreement with Buyer;

- (w) A Seller transfer or sale of the Project to another entity will be subject to Buyer's consent and the transferee must (i) accept all of Seller's obligations under the power purchase agreement between Buyer and Seller, including all duties, liabilities and indemnities, and (ii) either enter into new CSAs containing the same terms and conditions as the original CSAs with existing Customers, or accept assignment of the existing CSAs with existing Customers. In addition, Seller shall provide Customers with notice of any such transfer or sale of the Project;
- (x) Seller shall notify Customers of any proposed modifications to the Project and provide Customers adequate time to withdraw their Subscription to the Project due to any such proposed modifications;
- (y) A Customer's minimum Subscription must be projected to be an amount of energy per year equal to or greater than: (x) 100 kWh per month on average, calculated on an annual basis or (y) twenty five percent (25%) of such Customer's load, or as otherwise required by Green-e® Energy;
- (z) Within sixty (60) days after the Commercial Operation Date, Seller must provide completed Disclosure Documents and a statement that Seller is required by Green-e® Energy to provide updated Disclosure Documents to Customer on an annual basis;
- (aa) Seller will not make any statements or representations in the CSA or its marketing materials implying that renewable energy is being used or delivered to anyone unless Seller knows that Renewable Energy Credit ownership supports such statements;
- (bb) Seller representation that any electricity, stripped of Renewable Energy Credits is null power and no longer renewable and that, due to change of law provisions in the power purchase agreement between Buyer and Seller, power delivered may cease to be renewable;
- (cc) Seller covenants not to claim the Renewable Energy Credits associated with any Metered Amounts;
- (dd) Seller obligation regarding transfer and chain of custody of Renewable Energy Credits;
- (ee) Seller shall provide Customer notice of any direct change of control of Seller (whether voluntary or by operation of law);
- (ff) Seller shall disclose to Customers whether or not Seller will pursue Full Capacity Deliverability Status for the Project and the effects of achieving or not achieving Full Capacity Deliverability Status on the amount Customers will receive in bill credits;
- (gg) Customer acknowledgment that rates offered under the CSA may not ultimately be cheaper than those offered under utility service and that the

only aspect of service from Seller that can be kept constant for the term of the CSA is the generation rate component of the Customer's bill; and

(hh) The generation rate component shall be a term in the CSA.

3.36 <u>Distributed Energy Resource Aggregation.</u>

Throughout the Delivery Term, Seller covenants that it shall: (i) be a Distributed Energy Resource Provider, (ii) have a fully executed Distributed Energy Resource Provider Agreement; (iii) aggregate each of the Distributed Energy Resource(s) to a single Distributed Energy Resource Aggregation; (iv) promptly provide Notice to SCE if there are any modifications to Seller's Master File, and such Notice shall identify the exact modification(s), the reason for the modification(s), and Seller shall provide supporting documentation as reasonably requested by SCE; and (v) promptly provide Notice to SCE if there are any modifications to Seller's Generation Distribution Factor, and such Notice shall identify the exact modification(s), the reason for the modification(s), and Seller shall provide supporting documentation as reasonably requested by SCE.

3.37 FERC Qualifying Facility Status.

Seller shall take all actions, including making or supporting timely filings with FERC necessary to obtain or maintain the Qualifying Facility status of the Generating Facility throughout the Delivery Term; *provided, however*, that Seller shall not be obligated under this Section 3.37 to take any actions or make any filings to the extent that no action or filing is required by FERC to obtain, or maintain the Qualifying Facility status of the Generating Facility.

3.38 Common PNode Distributed Energy Resource(s).

Commencing on the Effective Date and continuing throughout the Term, Seller represents, warrants, and covenants that each Distributed Energy Resource is electrically connected to, or otherwise associated with, a single common PNode."

36. Section 4.01(a) is deleted and replaced with the following:

"SCE shall provide information to Seller regarding CAISO Revenues, CAISO Costs, CAISO Sanctions, and SCE Penalties and shall pay Seller and issue bill credits to Customers for Product in accordance with: (i) approved, accurate, and undisputed Subscription Information and Bill Credit Instructions, and (ii) Exhibit E.".

37. Section 4.01(b) is deleted and replaced with the following:

"(b) Throughout the Delivery Term, SCE shall purchase Product generated by the Generating Facility and delivered at the Delivery Point in accordance with this

Agreement, CAISO Tariff and Applicable Law, provided, subject to Section 4.01(c), SCE has no obligation to purchase from Seller or issue bill credits to Customers any Product that is not or cannot be delivered to the Delivery Point as a result of any circumstance, including:

- (i) An outage of the Generating Facility;
- (ii) A Force Majeure under Article Five; [or] {SCE Comment: For Full Buy-Sell DERs}
- (iii) A reduction or curtailment of deliveries in accordance with Section 3.12(g), except as set forth in Section 4.01(c)[.] {SCE Comment: For Full Buy-Sells DERs} [;]{SCE Comment: For DERs with Excess-Sales}
- (iv) An increase in the Site Host Load. {SCE Comment: For Full Buy-Sell DERs}.".
- 38. Section 4.01(c) is amended by adding the phrase "or issue bill credits to Customers" after the phrase "obligated to pay Seller" and before the words "for any CP".
- 39. [Intentionally Omitted.]
- 40. Section 4.01(e) is amended by adding the phrase "or issue bill credits to Customers" after the phrase "SCE will not be obligated to pay Seller" and before the words "for any Product".
- 41. New Sections 4.01(f), (g), and (h) are added as follows:
 - "(f) Buyer shall make payment of each invoice related to the Unsubscribed Capacity directly to Seller and in accordance with Exhibit E. Buyer and Seller acknowledge that payment to Seller under this Agreement of each invoice related to the Subscribed Delivered Energy shall be made by Buyer in the form of bill credits to Customers in accordance with the Seller's Subscription Information and Bill Credit Instructions, the GTSR-CR Decisions and Exhibit E, and Seller hereby assigns any right to receive all such payments in respect of Subscribed Delivered Energy to such Customers.
 - (g) Notwithstanding any other provision in this Agreement, Buyer is not obligated to provide a bill credit: (i) to any Customer that does not meet the requirements of this Agreement, or (ii) if Buyer determines, in its reasonable discretion, that the information contained in the Subscription Information and Bill Credit Instructions is incorrect. Retroactive changes to Subscription Information and Bill Credit Instructions will not be permitted.

- (h) Seller shall pay all CAISO charges, CAISO Sanctions, and CAISO Costs, and SCE shall retain all CAISO Revenues resulting from such violation of Section 1.05(d).".
- 42. Section 4.02(c)(ii) is delete and replaced with the following:
 - "(ii) If the deviation: (x) between the CAISO Approved Meter data and the Check Meter data for any comparison is greater than 0.3%, or (y) between the CAISO Approved DER Meter data and the Check Meter data, for each Distributed Energy Resource, for any comparison is greater than 0.3%, then SCE shall provide Notice to Seller of such deviation and the Parties will mutually arrange for a meter check or recertification of the Check Meter or CAISO Approved Meter, as applicable."
- 43. Section 4.03 is deleted and replaced with the following:
 - "4.03 Scheduling Coordinator.

Commencing on the Commercial Operation Date, SCE, or its designee, shall act as Seller's Scheduling Coordinator and carry out all duties as Scheduling Coordinator in accordance with CAISO Tariff protocols.".

- 44. Section 4.05(a) is amended by adding the following sentence at the end thereof: "Notwithstanding the foregoing, Buyer shall retire Renewable Energy Credits associated with Subscribed Delivered Energy on behalf of Subscribed Customers.".
- 45. Sections 6.01(b)(iv), (ix), (x), (xiii), (xiv), (xv), and (xx) are deleted and replaced with the following:
 - "(iv) Except as permitted in Sections 10.04 and 10.05, Seller does not own or otherwise have control of the entire Generating Facility;
 - (ix) Seller installs generating capacity in excess of the Contract Capacity or in excess of each DER Contract Capacity at each applicable DER Site, and such excess generating capacity is not removed within five (5) Business Days after Notice from SCE; {SCE Comment: Intermittent only}
 - (x) Seller installs direct current electric energy generating capacity in excess of the Installed DC Rating or in excess of each DER Installed DC Rating at each applicable DER Site and such excess direct current energy generating capacity is not removed within five (5) Business Days after Notice from SCE; {SCE Comment: For Solar Photovoltaic.}
 - (xiii) The entire Generating Facility fails to qualify as an ERR;
 - (xiv) Any electric energy from the Generating Facility and sold or to be sold to SCE hereunder fails to qualify as eligible renewable energy for purposes of the RPS Legislation;

- (xv) A termination of, or cessation of service under, any agreement necessary for Seller:
 - (1) To interconnect the Generating Facility to the Transmission Provider's electric system;
 - (2) To transmit the electric energy on the Transmission Provider's electric system;
 - (3) To comply with the CAISO Tariff; or
 - (4) To connect each Distributed Energy Resource to the Distributed Energy Resource Aggregation;

provided, if SCE and Seller mutually agree that a termination of, or cessation of service under, any such agreement is not due to the fault of Seller, Seller shall have thirty (30) days from such termination or cessation to cure such default:

- (xx) Seller transfers or assigns any Interconnection Queue Position or any interconnection agreement without the written consent of SCE.".
- 46. New Sections 6.01(b)(xxviii) and (xxix) are added as followings:
 - "(xxviii) Seller fails to maintain its status as a Distributed Energy Resource Provider; or
 - (xxix) Subject to Section 3.36, the Generating Facility or any DER fails to maintain its status as a Qualifying Facility.".
- 47. Section 10.02(a)(i) is amended to add the phrase "or control" after the phrase "Seller shall own".
- 48. Section 10.02(d) is amended to add the following new sentence at the end thereof:
 - "The last sentences in Sections 10.02(b) and 10.02(c) shall not be applicable to Seller's representations, warranties and covenants in the remaining portions of Sections 10.02(b) and 10.02(c). If Seller breaches or fails to perform its representations, warranties and covenants under Sections 10.02(b) and 10.02(c), such breach or failure to perform and satisfy the obligations under such sections shall be considered an Event of Default by Seller."
- 49. New Sections 10.02(h), (i), (j) and (k) are added as follows:
 - "(h) Seller and, if applicable, its successors, represents, warrants and covenants that prior to the Effective Date, on the Effective Date and continuing throughout the Delivery Term:

- (i) Seller has received from Buyer and has read Attachment 1 of the CPUC's CCA Code of Conduct decision (D.12-12-036) and has not and will not circumvent it;
- (ii) Seller has and shall continue to incorporate in each CSA it enters into with Customers the provisions required to be included in the CSA as identified in Sections 3.32 and 3.35;
- (iii) Seller has not and shall continue not to use Buyer's corporate name, trademark, trade name, logo, identity or any affiliation for any reason without Buyer's prior written consent; and
- (iv) Seller has not and will not enter into CSAs for Subscribed Capacity exceeding, in the aggregate, one hundred percent (100%) of the Contract Capacity; and Seller has not and will not enter into a CSA with any individual Customer for a Subscription exceeding 2 MW (except in the case of federal, state or local governments, schools or school districts, county offices of education, any of the California Community Colleges, the California State University or the University of California).
- (i) Seller and, if applicable, its successors, represents, warrants and covenants that throughout the Delivery Term:
 - (i) [Intentionally Omitted.];
 - (ii) The Metered Amounts qualify as Green-e® Energy eligible and are Green-e® Energy certified product;
 - (iii) The Subscription Information and Bill Credit Instructions required under Section 3.30 shall be accurate and complete; and
 - (iv) The Project complies with the requirements of the California Air Resources Board's Voluntary Renewable Electricity Program.
 - (v) Seller shall comply with the Residential Requirement, and if at any point in time during the Delivery Term Seller does not have sufficient Residential Customer Subscriptions to comply with the Residential Requirement, Seller shall cease to Subscribe non-residential Customers until there are sufficient Residential Customer Subscriptions to comply with the Residential Requirement.
- (j) Seller acknowledges that the Subscriptions it sells may be considered securities under federal or California law and, accordingly, Seller represents that it has retained its own legal counsel to provide advice on securities law matters.
- (k) With respect to the legal opinion delivered pursuant to Section 3.12(c)(xiv), Seller hereby represents and covenants that:
 - (i) The lawyer primarily responsible for the issuance of the opinion has, within the last eight (8) years, practiced federal and California securities

- law as a significant portion of their practice (meaning at least five (5) full-time years), and such experience included registering or qualifying offerings or sales of securities, effecting private placements of securities, and/or advising issuers or sellers of securities with respect to exemptions from qualification and registration requirements;
- (ii) The lawyer primarily responsible for issuance of the opinion is licensed to practice law in California and the lawyer's license is active and not under suspension; and
- (iii) The law firm issuing the opinion carries a minimum of ten million dollars (\$10,000,000) in professional liability insurance coverage that includes coverage for securities practice.".
- 50. A new Section 10.03(b)(viii) is added as follows:
 - "(viii) Seller shall defend, hold harmless and indemnify Buyer and its parent company, subsidiaries, affiliates, and its and their directors, officers, employees, shareholders, successors, and assigns from any and all damages, losses, or liability (including reasonable attorney's fees) for any and all claims or causes of action arising from or in connection with Seller's Subscription Information and Bill Credit Instructions, subscriptions, bill credits, disputes, violations of Applicable Laws, misrepresentations made by Seller or Seller's contractors, agents, or representatives, claims relating to securities laws, or Green-e® Energy certification, or loss thereof."
- 51. Section 10.04(a) is amended by adding the following new sentence at the end thereof: "In the case of an assignment of this Agreement by Seller, the assignee must assume the rights and obligations of the Seller under each CSA.".
- 52. Section 10.09(d) is amended by adding the following new sentence at the end thereof: "No Customer or any other third party shall be a third party beneficiary of this Agreement.".
- 53. A new section 10.09(r) is added as follows:
 - "(r) Nothing contained in this Agreement shall be construed as creating any relationship whatsoever between Buyer and Seller, including that of partners, coemployment, or joint venture parties.".
- 54. Section 10.10(a)(vii) is amended by adding the words "Green-e® Energy," after the words "CPUC, CEC, FERC,".
- 55. Section 10.14 is deleted and replaced with the following:

- "Except as specifically provided in this Agreement, any outstanding and past due payment amounts owing and unpaid by either Party under the terms of this Agreement, excluding bill credits to Customers, will be eligible to receive a Late Payment Simple Interest calculated using the Interest Rate for the number of days between the date due and the date paid."
- 56. Section 10.15 is amended by adding the phrase ", issuing bill credits to Customers," after the phrase "made by wire transfer".
- 57. Exhibit A is amended to delete the following defined terms:
 - "Accepted Compliance Costs",
 - "Compliance Actions",
 - "Compliance Expenditure Cap", and
 - "TOD Period Product Payment".
- 58. The definition of "CAISO Approved Meter" in Exhibit A is deleted and replaced with the following: "CAISO Approved Meter" means a CAISO approved revenue quality meter or meters, CAISO approved data processing gateway or remote intelligence gateway, telemetering equipment and data acquisition services sufficient for monitoring, recording and reporting, in real time, all electric energy produced by the Generating Facility less Station Use [and Site Host Load] {SCE Comment: For DERs with Excess-Sales only}.
- 59. The definition of "Curtailed Product Payment" in Exhibit A is amended by inserting the words "or bill credits" after the phrase "the sum of all payments".
- 60. The definition of "Demonstrated Contract Capacity" is deleted and replaced with the following:
 - "Demonstrated Contract Capacity" means the Generating Facility's total rated electric alternating current energy generating capacity, which will equal the sum of each DER Demonstrated Contract Capacity.
- 61. The definition of "Demonstrated Installed DC Rating" is deleted and replaced with the following:
 - "Demonstrated Installed DC Rating" means the sum of each DER Demonstrated Installed DC Rating.
 - {SCE Comment: DER Solar Photovoltaic only.}
- 62. The definition of "Generating Facility" in Exhibit A is deleted and replaced with the following:

- "Generating Facility" means all of Seller's Distributed Energy Resource(s), which when taken together constitute Seller's Distributed Energy Resource Aggregation.
- 63. The definition of "Lost Output" in Exhibit A is deleted and replaced with the following:
 - "Lost Output" means the reduction in Qualified Amounts over the relevant measurement period that the Generating Facility was available to produce and could reasonably have been expected to deliver, as reasonably determined by SCE, but was not delivered due to a Lost Output Event.
- 64. The definition of "Operate", "Operated", "Operating" or "Operation" in Exhibit A is deleted and replaced with the following:
 - "Operate", "Operated", "Operating" or "Operation" means to provide (or the provision of) all the operation, engineering, purchasing, repair, supervision, training, inspection, testing, protection, use, management, improvement, replacement, refurbishment, retirement, and maintenance activities associated with operating each Distributed Energy Resource in accordance with Prudent Electrical Practices.
- 65. The definition of "Paid Curtailed Product" in Exhibit A is amended by inserting the words "or provide bill credits to Customer(s)," after the phrase "SCE is obligated to pay Seller".
- 66. The definition of "Product Price" in Exhibit A is deleted and replaced with the following:
 - "Product Price" means the Subscribed Product Price or the Unsubscribed Product Price, as applicable.
- 67. The definition of "Site" is deleted and replaced with the following:
 - "Site" means the real property on which the Distributed Energy Resources constituting the Generating Facility are, or will be located, as further described in Section 1.01(b) and Exhibit B [, but excluding (a) that portion on which the Other Generating Facility is, or will be, located, as further described in Exhibit B, and (b) the Shared Facilities Area]. [SCE Comment: only applicable to projects that utilize Shared Facilities.]
- 68. The definition of "Station Use" is deleted and replaced with the following:
 - "Station Use" means:
 - (a) The electric energy produced by each Distributed Energy Resource that is used within each Distributed Energy Resource to power the lights, motors, control systems and other electrical loads that are necessary for Operation; and

- (b) The electric energy produced by each Distributed Energy Resource that is consumed within each Distributed Energy Resource electric energy distribution system as losses.
- 69. The definition of "Telemetry System" is deleted and replaced with the following:
 - "Telemetry System" means a system of electronic components that interconnects the Distributed Energy Resource Aggregation, GMS and the CAISO as set forth in Section 3.08(e).
- 70. The definition of "Term Year" in Exhibit A is deleted and replaced with the following: "Term Year" means a twelve (12) month period beginning on the Commercial Operation Date and each successive twelve (12) month period thereafter.
- 71. The definition of "Transmission Provider" is deleted and replaced with the following:
 - "Transmission Provider" means any entity or entities responsible for the interconnection of each Distributed Energy Resource with a Control Area or transmitting the Metered Amounts on behalf of Seller from each Distributed Energy Resource to the Delivery Point.
- 72. Exhibit A is amended to add the following new defined terms:
 - "CAISO Approved DER Meter" means: (i) the meter that (a) is dedicated solely to a particular Distributed Energy Resource, (b) provides Settlement Quality Meter Data of such Distributed Energy Resource, (c) can be aggregated to the level of the Distributed Energy Resource Aggregation, and (d) which complies with any CAISO requirements, if applicable; and (ii) the data processing gateway or remote intelligence gateway, telemetering equipment and data acquisition services sufficient for monitoring, recording and reporting, in real time, all electric energy produced by each Distributed Energy Resource, less Station Use [and Site Host Load] {SCE Comment: for DERs with Excess-Sales only}.
 - "California Air Resources Board's Voluntary Renewable Electricity Program" means the voluntary program to reduce GHG emissions under the California Cap-and-Trade Program as set forth in 17 C.C.R. Sections 95841 and 95831.
 - "CCA Code of Conduct" means the code of conduct adopted by the CPUC related to interactions with community choice aggregators, pursuant to Senate Bill 790 and as set forth in decision D.12-12-036.
 - "CI Eligible Customer" means a customer of Buyer who takes bundled service from Buyer, including having all its power requirements purchased by Buyer, and whose

service account is located within the same county as the Project or within ten (10) miles of the Project.

"Commitment(s) to Enroll" has the meaning set forth in Exhibit R-1.

"Community Interest Requirement" means the requirement adopted in the GTSR-CR Decisions whereby Seller must demonstrate that members of the community where the Project is located are interested in, and supportive of, the Project. The Community Interest Requirement can be satisfied with documentation that (i) CI Eligible Customers have (a) completed Commitments to Enroll in thirty percent (30%) of the Contract Capacity; or (b) completed Expressions of Interest for fifty-one percent (51%) of the Contract Capacity; and (ii) the Project has a minimum of three separate CI Eligible Customers who have completed Commitments to Enroll or Expressions of Interest, as applicable. The Community Interest Requirement can also be satisfied with a guarantee from the municipality where the Project is located that (i) CI Eligible Customers have committed to Subscribe to at least thirty percent (30%) of the Contract Capacity; or (ii) CI Eligible Customers have provided Expressions of Interest sufficient to reach a Subscription rate of fifty-one percent (51%) of the Contract Capacity.

"CR" has the meaning set forth in the preamble.

"CR Tariff" means that tariff available to customers of Buyer, between Buyer and customer, such that customers may become a Customer of Seller's Generating Facility.

"Customer" means a customer of Buyer who takes bundled services from Buyer including having all its power requirements purchased by Buyer, and who has signed up under the CR Tariff to receive benefits from Seller's Generating Facility.

"Customer-Seller Agreement" or "CSA" means that agreement to be executed between Customer and Seller in order for Customer to Subscribe to Seller's Generating Facility, which shall be subject to those requirements set forth within Section 3.34 of this Agreement. Buyer shall not be a party to the Customer-Seller Agreement.

"Default Load Aggregation Point" or "DLAP" has the meaning set forth in the CAISO Tariff.

"Default Load Aggregation Point Price" or "DLAP Price" means the hourly Integrated Forward Market Default Load Aggregation Point Locational Marginal Price as determined by the CAISO for the Buyer's applicable CAISO Transmission Access Charge Area.

"DER Contract Capacity" means for each DER Site, the lesser of (i) the amount of electric energy generating capacity, set forth in numeric order in Section 1.01(h), that

Seller commits to install at such DER Site and (ii) the DER Demonstrated Contract Capacity for such DER Site.

"DER Demonstrated Contract Capacity" means the total rated electric alternating current energy generating capacity of a particular Distributed Energy Resource

, which will equal the sum of the Inverter Block Unit Capacity of all Inverter Block Units of such Distributed Energy Resource, as determined in accordance with Exhibit J. {SCE Comment: DERs that are Solar Photovoltaic only.}

, which will equal the sum of the manufacturer's nameplate ratings of all installed Wind Turbines of such Distributed Energy Resource, consistent with Prudent Electrical Practices and accepted industry standards, as indicated on the nameplates physically attached to the individual Wind Turbine generators, as determined in accordance with Exhibit J.

{SCE Comment: DERs that are Wind only.}

, which will equal the sum of the Metered Amounts attributable to such Distributed Energy Resource for the Demonstration Hour, as determined in accordance with Exhibit J.

{SCE Comment: All other DER technologies.}

"DER Demonstrated Installed DC Rating" means the sum of the Photovoltaic Module DC Ratings for all Photovoltaic Modules of a particular Distributed Energy Resource actually installed at the applicable DER Site and verified by SCE pursuant to Exhibit J. {SCE Comment: DERs that are Solar Photovoltaic only.}

"DER Installed DC Rating" means for each DER Site, the lesser of (i) the amount of direct current electric energy generating capacity, set forth in numeric order in Section 1.01(i), that Seller commits to install at such DER Site, and (ii) the DER Demonstrated Installed DC Rating, expressed in kWpdc associated with such DER Site. {SCE Comment: For DERs that are Solar Photovoltaic only}.

"DER Site" means the real property on which a particular Distributed Energy Resource is, or will be located, as further described in numeric order in Section 1.01(b) and Exhibit B.

{SCE Comment: For DERs only}.

"DER Telemetry System" means a system of electronic components that interconnects each Distributed Energy Resource to the Distributed Energy Resource Aggregation, as set forth in Section 3.08(e).

{SCE Comment: For DERs only}.

"Disclosure Documents" means those disclosure documents required by Green-e® Energy to be provided by Seller to Customers and potential Customers, as they may be

amended, supplemented or replaced from time to time, as set forth on the Green-e® Energy website at http://green-e.org/verif docs.html, or successor thereof.

"Distributed Energy Resource(s)" or "DER(s)" means each of Seller's newly constructed electric generating facilities, which have never generated electricity before the Commercial Operation Date, except for testing pursuant to this Agreement, each as more particularly described in Exhibit B, together with all materials, equipment systems, structures, features and improvements necessary to produce electric energy at each facility, [and, with respect to the Shared Facilities, Seller's interests in such Shared Facilities] excluding each DER Site, land rights and interests in land, and as further defined in the CAISO Tariff. {SCE Comment: Bracketed language only applicable to projects that have Shared Facilities}.

"Distributed Energy Resource Aggregation" means the aggregation of each of Seller's Distributed Energy Resources, and as further described in the CAISO Tariff.

"Distributed Energy Resource Provider" has the meaning set forth in the CAISO Tariff.

"Distributed Energy Resource Provider Agreement" has the meaning set forth in the CAISO Tariff.

"Expected Annual Site Host Load" has the meaning set forth in Section 1.01(i). {SCE Comment: DERs with Excess-Sales Only.}

"Expression(s) of Interest" has the meaning set forth in Exhibit R-1.

"FTC" means the Federal Trade Commission.

"FTC Green Guides" means those guiding documents published on the FTC website intended to provide guidance on (1) general principles applicable to environmental marketing claims, (2) how consumers are likely to interpret particular claims and how marketers can substantiate these claims, and (3) how marketers can qualify their claims to avoid deceiving customers.

"Generation Distribution Factor" has the meaning as set forth in the CAISO Tariff.

"Green-e® Energy" means the national certification program for renewable energy administered by the Center for Resource Solutions, as such program may be amended, supplemented or otherwise changed from time to time, and about which information can be found at http://www.green-e.org, or successor thereof.

"Green-e® Energy Code of Conduct" means the code of conduct published by Green-e® Energy that outlines the requirements for consumer protection and environmental standards, and which can be found at http://www.green-e.org, or successor thereof.

- "GTSR" has the meaning set forth in the preamble.
- "GTSR-CR Decisions" means the CPUC Decisions promulgated pursuant to Senate Bill 43, including Decisions 15-01-051, 16-05-006, and any other existing or future ruling, decision, or regulation related to GTSR enacted, adopted or promulgated by any applicable Governmental Authority.
- "GTSR CR Rider" has the meaning set forth in the preamble.
- "Integrated Forward Market" has the meaning as set forth in the CAISO Tariff.
- "Master File" has the meaning set forth in the CAISO Tariff.
- "Minimum Subscription Requirement" has the meaning set forth in Exhibit E.
- "PURPA" means the Public Utility Regulatory Policies Act of 1978, Public Law, 95617, as amended from time to time.
- "Qualifying Facility" means an electric energy generating facility that complies with the qualifying facility definition established by PURPA and any FERC rules as amended from time to time (18 Code of Federal Regulations Part 292, Section 292.203 et seq.) implementing PURPA and, to the extent required to obtain or maintain Qualifying Facility status, is self-certified as a Qualifying Facility or is certified as a Qualified Facility by FERC.
- "Renewable Energy Credit Market Price" means \$10/MWh.
- "Residential Customer" means a domestic class of Customer whose dwelling is a single-family unit, multi-family unit, mobilehome or similar living establishment.
- "Residential Requirement" means the requirement adopted in the GTSR-CR Decisions that 1/6 of the Subscriptions, in kW, must come from Residential Customers.
- "Settlement Quality Meter Data" has the meaning set forth in the CAISO Tariff.
- "Site Host" means the person, persons, or other entity purchasing or otherwise using the Site Host Load or thermal energy output from each Distributed Energy Resource. {SCE Comment: For DERs with Excess-Sales only.}
- "Site Host Load" means the electric energy produced by or associated with each Distributed Energy Resource that serves electrical loads (that are not Station Use) of Seller or one or more third parties conducted pursuant to California Public Utilities Code Section 218(b).
- {SCE Comment: For DERs with Excess-Sales only.}

"Subscribed Capacity" means the aggregate Subscription level of all Customers with Subscriptions to the Generating Facility for each month.

"Subscribed Delivered Energy" means that portion of the Qualified Amounts equal to the product of (a) the sum of Qualified Amounts minus (i) any electric energy produced by the Generating Facility for which SCE is not obligated to pay Seller as set forth in Section 4.01(e), in kWh, and minus (ii) any electric energy produced by the Generating Facility as set forth in Section 1.05(c)(ii), in kWh, in all hours for the TOD Period being calculated, measured in kWh; and (b) the quotient of the Subscribed Capacity divided by the Contract Capacity.

"Subscribed Product Price" has the meaning set forth in Section 1.05.

"Subscription", "Subscribe", "Subscribed" and other grammatical variations thereof means:

- (a) In the case of a capacity-based subscription business model employed in the CSA, the subscription that a Customer has signed up for, expressed in kW.
- (b) In the case of an energy-based subscription business model employed in the CSA, the monthly subscription that a Customer has signed up for (expressed in kWh/month), multiplied by the Contract Capacity (expressed in kW), divided by the Expected Annual Net Energy Production (expressed in kWh/year), multiplied by 12 months/year, the product of which shall be equal to the Subscription of the Customer, expressed in kW.

Example: Customer energy subscription (kWh/month) x Contract Capacity (kW) x 12 (months/year) / Expected Annual Net Energy Production (kWh/year) = Subscription (kW).

"Subscription Information and Bill Credit Instructions" means the information required to be provided by Seller to Buyer in accordance with Section 3.29 and as set forth in the form provided in Exhibit P.

"Unsubscribed Capacity" means the Contract Capacity minus the Subscribed Capacity for each billing month.

"Unsubscribed Delivered Energy" means that portion of the Qualified Amounts equal to the product of (a) the sum of Qualified Amounts, in all hours for the TOD Period being calculated, measured in kWh, and (b) the quotient of the Unsubscribed Capacity divided by the Contract Capacity.

"Unsubscribed Product Price" has the meaning set forth in Section 1.05.

73. Section 2(f) of Exhibit D is amended to delete the first paragraph and replace it with the following:

If Seller is Forecasting electric energy, in accordance with SCE's instructions, and Seller learns of any change in the total electrical energy output of the Generating Facility for a period covered by the most recent Forecast update resulting from any cause, including an unplanned outage, before the time that the next weekly update of the Forecast is due which results in variance in expected energy in any hour of plus (+) or minus (-) three percent (3%) from the energy reported in the most recent Forecast update, Seller shall provide an updated Forecast to SCE. This updated Forecast must be submitted to SCE by no later than:

{SCE Comment: For Full Buy-Sell only.}

If Seller is Forecasting electric energy, in accordance with SCE's instructions, and Seller learns of any change in the excess electrical energy output of the Generating Facility for a period covered by the most recent Forecast update resulting from any cause, including an unplanned outage or Site Host Load changes, before the time that the next weekly update of the Forecast is due which results in variance in expected energy in any hour of plus (+) or minus (-) three percent (3%) from the energy reported in the most recent Forecast update, Seller shall provide an updated Forecast to SCE. This updated Forecast must be submitted to SCE by no later than:

{SCE Comment: For DER with Excess-Sales only.}

- 74. Exhibit E is deleted in its entirety and replaced with a new Exhibit E, as attached hereto.
- 75. Exhibit J-1 and Exhibit J-2 are deleted and replaced with the Exhibit J-1 and Exhibit J-2 respectively, attached hereto.
- 76. Exhibit K-1, Exhibit K-2, Exhibit K-3, and Exhibit K-4 are deleted and replaced with a new Exhibit K, as attached hereto.
- 77. Exhibit O-1, Exhibit O-2a, and Exhibit O-2b are each amended to delete the first sentence and replace it with the following:
 - "Pursuant to Section 3.08(f), Seller shall install and maintain a minimum of one (1) stand-alone meteorological equipment station for DER Site or for each one (1) square mile (or portion thereof) of each DER Site.".
- 78. A new Exhibit P, as attached hereto, is added to the exhibits after Exhibit O.
- 79. A new Exhibit Q-1, as attached hereto, is added to the exhibits after Exhibit P.

- 80. A new Exhibit Q-2, as attached hereto, is added to the exhibits after Exhibit Q-1.
- 81. A new Exhibit R-1, as attached hereto, is added to the exhibits after Exhibit Q-2.
- 82. A new Exhibit R-2, as attached hereto, is added to the exhibits after Exhibit R-1.
- 83. A new Exhibit R-3, as attached hereto, is added to the exhibits after Exhibit R-2.

84. MISCELLANEOUS

- (a) <u>Reservation of Rights</u>. Each of the Parties expressly reserves all of its respective rights and remedies under the Agreement.
- (b) <u>Legal Effect</u>. Except as expressly modified as set forth herein, the Agreement remains unchanged and, as so modified, the Agreement shall remain in full force and effect. Each of the Parties hereby represents and warrants that the representations contained in the Agreement are true on and as of the date hereof as if made by the Party on and as of said date.
- (c) Governing Law. THIS GTSR CR RIDER AND THE RIGHTS AND DUTIES OF THE PARTIES HEREUNDER SHALL BE GOVERNED BY AND CONSTRUED, ENFORCED AND PERFORMED IN ACCORDANCE WITH THE LAWS OF THE STATE OF CALIFORNIA, WITHOUT REGARD TO PRINCIPLES OF CONFLICTS OF LAW. TO THE EXTENT ENFORCEABLE AT SUCH TIME, EACH PARTY WAIVES ITS RESPECTIVE RIGHT TO ANY JURY TRIAL WITH RESPECT TO ANY LITIGATION ARISING UNDER OR IN CONNECTION WITH THIS GTSR CR RIDER.
- (d) <u>Successors and Assigns</u>. This GTSR CR Rider shall be binding upon and inure to the benefit of the Parties hereto and their respective successors and assigns.
- (e) <u>Authorized Signatures; Notices</u>. Each Party represents and warrants that the person who signs below on behalf of that Party has authority to execute this GTSR CR Rider on behalf of such Party and to bind such Party to this GTSR CR Rider. Any written notice required to be given under the terms of this GTSR CR Rider shall be given in accordance with the terms of the Agreement.
- (f) <u>Effective Date</u>. This GTSR CR Rider shall be deemed effective as of the Execution Date (the "GTSR CR Rider Effective Date").
- (g) <u>Further Agreements</u>. This GTSR CR Rider shall not be amended, changed, modified, abrogated or superseded by a subsequent agreement unless such subsequent agreement is in the form of a written instrument signed by the Parties.

(h) <u>Counterparts; Electronic Signatures.</u> This GTSR CR Rider may be executed in one or more counterparts, each of which will be deemed to be an original of this GTSR CR Rider and all of which, when taken together, will be deemed to constitute one and the same agreement. The exchange of copies of this GTSR CR Rider and of signature pages by facsimile transmission, Portable Document Format (i.e., PDF), or by other electronic means shall constitute effective execution and delivery of this GTSR CR Rider as to the Parties and may be used in lieu of the original GTSR CR Rider for all purposes.

[Remainder of Page Left Intentionally Blank.]

IN WITNESS WHEREOF, the Parties hereto have caused this GTSR CR Rider to be executed as of the Execution Date.

[SELLER], a [State and form of incorporation].
By:
[Name] [Title]

SOUTHERN CALIFORNIA EDISON
COMPANY,
a California corporation.
By:
[Name]
[Title]

EXHIBIT E

Payments and Invoicing

1. COST RESPONSIBILITY, INVOICING AND PAYMENTS UPON COMMENCEMENT OF THE TERM

- 1.01 Cost Responsibility Upon Commercial Operation.
 - (a) <u>SCE Cost Responsibility</u>.

Upon the Commercial Operation Date and for the remainder of the Term,

- (i) Except under the circumstances set forth in Section 4.01(e), SCE shall make monthly Product Payments to Seller and issue bill credits to Customers for Product delivered to SCE calculated in the manner described in Section 1.02 below and Exhibit M;
- (ii) Except as set forth in Exhibit M and Section 1.01(b) of Exhibit E, SCE shall be responsible for all CAISO Costs and CAISO Sanctions and have the right to receive all CAISO Revenues;
- (iii) To the extent that SCE requires Seller to participate in the PIRP/EIRP program, SCE shall be responsible for PIRP/EIRP forecasting fees. {SCE Comment: For Intermittent Only.}
- (b) <u>Seller Cost Responsibility</u>.

Upon the Commercial Operation Date and for the remainder of the Term:

- (i) If Seller fails to comply with the Forecasting provisions set forth in Exhibit D, Seller shall pay an SCE Penalty as set forth in Exhibit M. {SCE Comment: Intermittent only.}
- (ii) Seller shall be responsible for CAISO Costs and CAISO Sanctions, under the circumstances specified in Section 4.01(e) [and RA Deficit Payments under the circumstances specified in Section 3.02] {SCE Comment: Only applicable if Seller has chosen the option to make RA Deficit Payments.}.
- (iii) Seller shall be responsible for Negative LMP Costs, if applicable, as set forth in Section 1.05(c)(i).
- (iv) Seller shall be responsible for CAISO Costs and CAISO Sanctions and shall receive CAISO Revenues under the circumstances specified in Section 1.05(c)(ii).
- (v) Seller shall make monthly payments calculated in the manner described in Section 1.02 below and Exhibit M.
- (vi) Seller shall be responsible for any and all CAISO charges, CAISO Costs and CAISO Sanctions under the circumstances specified in Section 1.05(d) and 4.01(h).

1.02 Product Payment Calculations After Commercial Operation Date.

For the purpose of calculating monthly payments and issuing bill credits to Customers for Product delivered to SCE as of the Commercial Operation Date in accordance with the terms of this Agreement ("Product Payments"), Qualified Amounts will be time-differentiated according to the TOD Periods set forth in Exhibit I, the pricing will be weighted by the Product Payment Allocation Factors set forth in Exhibit I, and:

- (i) if the quotient of the billing month's average Subscribed Capacity divided by the Contract Capacity is greater than the minimum Subscription level required for the corresponding billing month as specified in the table below ("Minimum Subscription Requirement"), then: (a) payment for Subscribed Delivered Energy shall be calculated pursuant to the GTSR-CR Decisions and shall be applied as a bill credit to Customers, and (b) payment for Unsubscribed Delivered Energy shall be calculated using the Subscribed Product Price, and shall be paid to Seller, net any amounts owed; and
- (ii) if the quotient of the billing month's average Subscribed Capacity divided by the Contract Capacity is less than the applicable Minimum Subscription Requirement, then: (a) payment for Subscribed Delivered Energy shall be calculated pursuant to the GTSR-CR Decisions and shall be applied as a bill credit to Customers, and (b) payment for Unsubscribed Delivered Energy shall be paid to Seller and shall be calculated using the lesser of the Unsubscribed Product Price and Subscribed Product Price, and shall be paid to Seller, net any amounts owed.

Years of Operation	Minimum subscription level for purposes of calculating the Minimum Subscription Requirement
First Term Year	45%
Second Term Year	70%
Third Term Year	90%
Fourth Term Year through the remainder of the Delivery Term	95%

Monthly Product Payments will equal the sum of (i) the sum of the monthly payments for all TOD Periods in the month and (ii) the Curtailed Product Payment. Each monthly

payment will be calculated pursuant to the following formulas, where "n" is the TOD Period being calculated:

SELLER'S PAYMENT_n (if Minimum Subscription Requirement is met) = $A \times C \times (D - E - F) \times [(I - H)/I] + G$

SELLER'S PAYMENT_n (if Minimum Subscription Requirement is not met) = B x (D - E - F) x [(I - H)/I] + G

Where:

A = Subscribed Product Price, in \$/kWh (i.e., \$/MWh/1000).

B = The Unsubscribed Product Price in \$\frac{1}{2}kWh (i.e., \frac{1}{2}MWh/1000).

C = Product Payment Allocation Factor for the TOD Period "n" being calculated.

D = The sum of Qualified Amounts in all hours for the TOD Period "n" being calculated in kWh.

E = Any electric energy produced by the Generating Facility for which SCE is not obligated to pay Seller as set forth in Section 4.01(e), in kWh.

F = Any electric energy produced by the Generating Facility as set forth in Section 1.05(c)(ii), in kWh.

G = The sum of all CAISO Revenues less the sum of all CAISO Costs and CAISO Sanctions, with respect to electric energy produced in all hours for the TOD Period "n" by the Generating Facility as set forth in Section 1.05(c)(ii).

H = Subscribed Capacity.

I = Contract Capacity.

Payments for Subscribed Delivered Energy shall be calculated pursuant to the GTSR-CR Decisions and shall be applied as a bill credit to Customers. Notwithstanding any other provision in this Agreement, in the event of any conflict or inconsistency between this GTSR CR Rider and the GTSR-CR Decisions regarding payment for Subscribed Delivered Energy as bill credits to Customers, the GTSR-CR Decisions shall control.

1.03 Payment During the Term.

For payments associated with Unsubscribed Delivered Energy, SCE shall, within ninety (90) days immediately following the end of each calendar month:

(a) Issue a Payment Invoice to Seller, and upon request from Seller, Buyer shall include documentation supporting any SCE Penalties, Negative LMP Costs,

- CAISO Costs, CAISO Sanctions, [RA Deficit Payments pursuant to Section 3.02] {SCE Comment: FCDS projects only.}, or other applicable revenues, charges and offsets which affected the net amount in the Payment Invoice; and
- (b) Send to Seller, via wire transfer or Automated Clearing House, SCE's payment of said net amount, less any applicable SCE Penalties, Negative LMP Costs, CAISO Costs, CAISO Sanctions, [RA Deficit Payments pursuant to Section 3.02] {SCE Comment: FCDS projects only.}, or other applicable charges or offsets plus, if such payment is late, a Late Payment Simple Interest calculated using the Interest Rate and the number of days that such payment is late.

Any amounts owed by Seller under this Agreement, including those set forth in Section 1.01(b) of this Exhibit E, shall not be included in Seller's Subscription Information and Bill Credit Instructions, but shall be included in amounts payable directly to or from Seller, and such amounts shall be paid in the form of bill credits and will not be eligible to receive a Late Payment Simple Interest if such bill credits are issued to Customer late, as a result of recomputation, error, or resolution of a disputed amount.

The Parties acknowledge that data necessary to calculate certain payment obligations of SCE and Seller under this Agreement may not be available at the time SCE issues the Payment Invoice with respect to a particular month. Any such payment obligations, shall be included in a subsequent Payment Invoice issued to Seller on or before the last Business Day of the month following the month that is the later of (i) one hundred and twenty (120) days following the last day of the calendar month to which the data relates or (ii) thirty (30) days after the relevant CAISO final settlement data is available to SCE.

1.04 Recomputation and Payment Adjustments.

- (a) If Seller or SCE determines that a calculation of Qualified Amounts, Metered Amounts, CAISO Revenues, Negative LMP Costs, CAISO Costs, CAISO Sanctions, [RA Deficit Payments] {SCE Comment: FCDS projects only.}, Subscribed Delivered Energy, Unsubscribed Delivered Energy, or SCE Penalties is incorrect as a result of inaccurate meters, the correction of data by the CAISO in OMAR, or a recalculation of CAISO Sanctions or other amounts owing between the Parties, Customer, Seller or SCE, as the case may be, shall promptly recompute the Qualified Amounts, Metered Amounts, CAISO Revenues, Negative LMP Costs, CAISO Costs, CAISO Sanctions, [RA Deficit Payments] {SCE Comment: FCDS projects only.}, SCE Penalties Subscribed Delivered Energy, Unsubscribed Delivered Energy, or other amounts for the period of the inaccuracy based upon an adjustment of inaccurate meter readings, correction of data or recalculation of CAISO Sanctions in accordance with the CAISO Tariff and any payment or bill credit affected by the adjustment or correction.
- (b) Adjustment of Payment.

Any amount due from SCE to Seller, or Seller to SCE, or any bill credit from SCE to Customer, as the case may be, will be made as an adjustment to the next

monthly Payment Invoice or bill credit that is calculated after Seller's or SCE's recomputation using corrected measurements.

If the recomputation results in a net amount owed to SCE after applying any amounts owing to Seller or bill credits to Customer as shown on the next monthly Payment Invoice, any such amount owing to SCE will at SCE's discretion be netted against amounts owed to Seller or against bill credits owed to Customer in any subsequent monthly Payment Invoice or bill credits or separately invoiced to Seller, in which case Seller must pay the amount owing to SCE within twenty (20) days after receipt of that invoice.

SCE may make payment adjustments arising from a recalculation of CAISO Revenues, CAISO Costs, Negative LMP Costs, CAISO Sanctions, [RA Deficit Payments] {SCE Comment: Only applicable if Seller has chosen the option to make RA Deficit Payments.}, SCE Penalties, Subscribed Delivered Energy, Unsubscribed Delivered Energy, or as a result of inaccurate meters after the end of the Term Year, provided, the Parties will be deemed to have waived any such payment adjustments which are not communicated as provided in this Section 1.04 of this Exhibit E within twelve (12) months after the end of the month in which the Payment Invoice was issued containing the error. Adjustment payments for meter inaccuracy will not bear interest.

1.05 <u>Netting Rights</u>.

SCE reserves the right to net amounts that would otherwise be due to Seller under this Agreement in payment of any amounts:

- (a) Owing and unpaid by Seller to SCE under this Agreement; or
- (b) Owed to SCE by Seller arising out of, or related to, any other SCE agreement, tariff, obligation or liability.

Nothing in this Section 1.05 of this Exhibit E limits SCE's rights under applicable tariffs, other agreements or Applicable Law.

1.06 Audit Rights.

Buyer, or its designee, shall have the right, at its sole expense and during normal working hours following Buyer's Notice, to audit the documents, records or data of Seller to the extent reasonably necessary to verify the accuracy of any statement, claim, charge or calculation made pursuant to this Agreement. Seller shall promptly comply with any Buyer Notice under this Section 1.06 of Exhibit E and provide copies of or access to documents, records or data to Buyer. The rights and obligations hereunder shall survive the termination of this Agreement for a period of five (5) years.

2. PAYMENT ERRORS

2.01 Notice of Error in Payment.

Except as provided in Section 1.04 of this Exhibit E, if within forty-five (45) days after receipt of SCE's Payment, Seller does not give SCE Notice of an error in the payment amount or bill credit, then Seller will be deemed to have waived any error in the payment.

2.02 Reimbursement for Underpayments and Overpayments.

If Seller identifies a payment error in Seller's favor and SCE agrees that the identified error occurred, SCE shall reimburse Seller or Customer for the amount of the underpayment caused by the error and apply the additional payment or bill credit to the next monthly Payment Invoice that is calculated.

If Seller identifies a payment error in SCE's favor and SCE agrees that the identified error occurred, SCE may net the amount of overpayment caused by the error against amounts otherwise owed to Seller or to Customer in connection with the next monthly Payment Invoice or bill credit that is calculated.

2.03 <u>Late Payments</u>.

Late payments to Seller resulting from SCE's errors, or overpayments to Seller by SCE, will include a Late Payment Simple Interest calculated using the Interest Rate and the number of days between the date due (or, in the case of overpayments by SCE, commencing five (5) Business Days from the date SCE provides Notice of such overpayments to Seller) and the date paid; *provided*, changes made because of settlement, audit or other information provided by the CAISO and not available to SCE when SCE rendered its original Payment Invoice will not bear interest.

2.04 Netting after Recomputation.

If the recomputation for an error results in a net amount still owing to SCE after applying the amounts owed to SCE against any amounts owed to Seller in the Payment Invoice or to Customers in bill credits, as described above, then SCE may, in its discretion, either net this net remaining amount owed to SCE against amounts owed to Seller in any subsequent monthly Payment Invoice to Seller or bill credits to Customers, or in the case of Seller invoice Seller for such amount, in which case Seller must pay the amount owing to SCE within twenty (20) days after receipt of such invoice.

2.05 <u>Resolution of Disputes</u>.

The Parties shall negotiate in good faith to resolve any disputes regarding claimed errors in a Payment. Any disputes which the Parties are unable to resolve through negotiation may be submitted for resolution through the mediation and arbitration as provided in Article Twelve. Upon resolution of the Dispute, any required payment or bill credit shall be made within ten (10) Business Days of such resolution, and in the case of a payment only shall include a Late Payment Simple Interest calculated using the Interest Rate from and including the due date but excluding the date paid.

*** End of Exhibit E ***

EXHIBIT J-1

Procedure for Demonstration of Contract Capacity

SCE Comment: For Intermittent only.

- 1. Seller's Notice of Demonstration Date.
 - Seller shall provide at least thirty (30) days prior Notice to SCE of the date selected by Seller ("Demonstration Date"), which Demonstration Date shall be no later than thirty (30) days following the Commercial Operation Date, during which Seller intends to demonstrate the Contract Capacity and the DER Contract Capacity. Upon SCE's request, Seller shall make reasonable efforts to reschedule the Demonstration Date.
- 2. <u>Demonstration of Contract Capacity, DER Contract Capacity, [Installed DC Rating, and DER Installed DC Rating] {SCE Comment: For Solar Photovoltaic DERs}.</u>
 - SCE shall complete a site visit on the Demonstration Date to verify that the Generating Facility was developed in accordance with the Generating Facility and Site Description set forth in Exhibit B and to determine the Demonstrated Contract Capacity, the DER Demonstrated Contract Capacity, [the Demonstrated Installed DC Rating, and the DER Demonstrated Installed DC Rating] {SCE Comment: For DERs that are Solar Photovoltaic}. In order to determine the Demonstrated Contract Capacity, SCE shall sum each of the DER Demonstrated Contract Capacities. In order to determine each DER Demonstrated Contract Capacity, SCE shall calculate the total nameplate rating for the generating equipment that is installed at each DER Site.

*** End of EXHIBIT J-1 ***

EXHIBIT J-2

Procedure for Demonstration of Contract Capacity

SCE Comment: For Baseload only.

1. Seller's Notice of Demonstration Hour.

Seller shall provide Notice to SCE of the date and hour selected by Seller, which hour must have occurred within thirty (30) days following the Commercial Operation Date, during which Seller claims it has demonstrated the applicable Contract Capacity and DER Contract Capacity ("Demonstration Hour").

- 2. Demonstration of Contract Capacity.
 - (a) Unless SCE provides timely Notice to Seller that additional days are required to substantiate data, SCE shall, within thirty (30) days after Seller's Notice of the Demonstration Hour, retrieve interval data downloaded from the CAISO Approved Meter or Check Meter for the twelve (12) hour periods before and after the Demonstration Hour; and
 - (b) SCE may, at its sole discretion, complete a site visit within thirty (30) days after SCE's receipt of Seller's Notice of the Demonstration Hour to verify that the Generating Facility and each Distributed Energy Resource was developed in accordance with the Generating Facility and Site Description set forth in Exhibit B.

*** End of EXHIBIT J-2 ***

EXHIBIT K-1

Lost Output

Lost Output, as used in Section 3.21, that is limited to Lost Output Events described in subsections (a), (b), or (d) of the definition of Lost Output Event, shall be estimated by Seller in accordance with the procedures described in this Exhibit K. Lost Output as a consequence of Lost Output Events described in Subsection (c) of the definition of Lost Output Event shall be determined by SCE in accordance with the definition of Curtailment Lost Output and shall not be included in the Lost Output Workbook defined in this Exhibit K.

Seller shall (i) collect the measurement data and perform the engineering calculations specified below in one (1) or more Microsoft Excel Workbooks (the "Lost Output Workbook") provided in a form and naming convention approved by SCE and (ii) electronically send the Lost Output Workbook to an address provided by SCE.

SCE shall have the right to verify all data by inspecting measurement instruments and reviewing Generating Facility Operating records.

Seller shall update the Lost Output Workbook each month and shall include the latest revision of the Lost Output Workbook with its monthly Lost Output Report.

1. <u>Log of Lost Output Events</u>.

The log of Lost Output Events must be created on a single, dedicated worksheet that is arranged with:

- (a) One (1) column for a unique Lost Output Event number;
- (b) One (1) column for the Term Year number;
- (c) One (1) column identifying the particular Distributed Energy Resource for which the event occurred;
- (d) One (1) column identifying the particular DER Site for which the event occurred;
- (e) One (1) column for the start date;
- (f) One (1) column for the start time;
- (g) One (1) column for the end date;
- (h) One (1) column for the end time;
- (i) One (1) column for the duration;
- (j) One (1) column for the cause;
- (k) One (1) column for the total of metered amounts associated with the Distributed Energy Resource for which the event occurred during all of the Settlement Intervals of the Lost Output Event;

- (l) One (1) column for the total of Seller's estimate of the Lost Output during all of the Settlement Intervals of the Lost Output Event;
- (m) One (1) column for a percentage calculated by *dividing* the preliminary results set forth in Item 1(l) of this Exhibit K by the Metered Amounts set forth in Item 1(k) of this Exhibit K; and
- (n) One (1) row for each Lost Output Event.

*** End of Exhibit K-1 ***

Exhibit P Subscribed Customer Reporting Form

Customer Subscription details are to be provided 60 days prior to the Commercial Operation Date, and afterwards, on a monthly basis to Buyer in the form attached below. Note that Seller should only fill out either the "Capacity Subscribed (kW)" or the "Load Subscribed (kWh)" column. The appropriate column shall be dictated by the business model being employed by Seller pursuant to the CSA.

Name	Service Address	SCE service account number	Capacity Subscribed (kW)	Load Subscribed (kWh)	Load Served (kW)

*** End of Exhibit P ***

EXHIBIT Q-1 QF Efficiency Monitoring Program – Cogeneration Data Reporting Form

		State:	Zip Code: _	
ID No.:	Genera	tion Nameplate (KW):		_
n Operation:Yes				
Can vour facility dun	np vour thermal out	put directly to the environment	t? □ _{Yes} □	No
Ownership	1 7	,		
Name		Address		nership (%) Utility —— Y N
				Y N
PrevYear] Monthly	Operating Data			
PrevYear] Monthly Indicate the BTUs If Energy In Useful Po	Operating Data unit of measure use	ed for your Useful Thermal End	ergy Output if BTUs LHV) as suppli	other than mBTUs:
PrevYear] Monthly Indicate the BTUs If Energy In Useful Po	Operating Data unit of measure use put is natural gas, u wer Output (1)	ed for your Useful Thermal End Therms mm use the Lower Heating Value (I	ergy Output if BTUs LHV) as suppli	other than mBTUs: ed by Gas Supplier. ful Thermal Energy Out
PrevYear] Monthly Indicate the BTUs If Energy In Useful Po (n	Operating Data unit of measure use put is natural gas, u wer Output (1)	ed for your Useful Thermal End Therms mm use the Lower Heating Value (I	ergy Output if BTUs LHV) as suppli	other than mBTUs: ed by Gas Supplier. ful Thermal Energy Out
PrevYear] Monthly Indicate the BTUs If Energy In Useful Po (n b ar	Operating Data unit of measure use put is natural gas, u wer Output (1)	ed for your Useful Thermal End Therms mm use the Lower Heating Value (I	ergy Output if BTUs LHV) as suppli	other than mBTUs: ed by Gas Supplier. ful Thermal Energy Out
PrevYear] Monthly Indicate the BTUs If Energy In Useful Po (n b ar or ay	Operating Data unit of measure use put is natural gas, u wer Output (1)	ed for your Useful Thermal End Therms mm use the Lower Heating Value (I	ergy Output if BTUs LHV) as suppli	other than mBTUs: ed by Gas Supplier. ful Thermal Energy Out
PrevYear] Monthly Indicate the BTUs If Energy In Useful Po (n b ar or ay n l	Operating Data unit of measure use put is natural gas, u wer Output (1)	ed for your Useful Thermal End Therms mm use the Lower Heating Value (I	ergy Output if BTUs LHV) as suppli	other than mBTUs: ed by Gas Supplier. ful Thermal Energy Out
PrevYear] Monthly Indicate the BTUs If Energy In Useful Po (n b ar or ay n I	Operating Data unit of measure use put is natural gas, u wer Output (1)	ed for your Useful Thermal End Therms mm use the Lower Heating Value (I	ergy Output if BTUs LHV) as suppli	other than mBTUs: ed by Gas Supplier. ful Thermal Energy Out
PrevYear] Monthly Indicate the BTUs If Energy In Useful Po (n b ar or ay n l ug p	Operating Data unit of measure use put is natural gas, u wer Output (1)	ed for your Useful Thermal End Therms mm use the Lower Heating Value (I	ergy Output if BTUs LHV) as suppli	other than mBTUs: ed by Gas Supplier. ful Thermal Energy Out
PrevYear] Monthly Indicate the BTUs If Energy In Useful Po (n b ar or ay n l ug p et	Operating Data unit of measure use put is natural gas, u wer Output (1)	ed for your Useful Thermal End Therms mm use the Lower Heating Value (I	ergy Output if BTUs LHV) as suppli	other than mBTUs: ed by Gas Supplier. ful Thermal Energy Out
PrevYear] Monthly Indicate the BTUs If Energy In Useful Po (n b ar or ay n l lg p et	Operating Data unit of measure use put is natural gas, u wer Output (1)	ed for your Useful Thermal End Therms mm use the Lower Heating Value (I	ergy Output if BTUs LHV) as suppli	other than mBTUs: ed by Gas Supplier. ful Thermal Energy Out

EXHIBIT Q-2 Fuel Use Standards – Small Power Producer Data Reporting Form

Street: City:		State:	Zip Code:	
Genera	ation Nameplate (KW):			
Prima	ry Energy: Biomass	s 🗌 Waste 🗎 Solar	☐ Other:	
	- 3 –			
Owner	rsnip		Ow	nership
N:	ame	Address	Ow	(%) Utility
		1 Tudi ess		•
				Y N
3				Y N
4				Y N
_				37 NT
				Y N
	Year] Monthly Operati	ng Data		
·	Year] Monthly Operati		Supplementary	Total Energy Inp
	Year] Monthly Operati	ng Data Primary Energy		
[Prev]	Year] Monthly Operati Useful Power Output (1)	ng Data Primary Energy Source (2)	Supplementary Energy Source (3)	Total Energy Inp (4)
[Prev]	Year] Monthly Operati Useful Power Output (1)	ng Data Primary Energy Source (2)	Supplementary Energy Source (3)	Total Energy Inp (4)
[Prev] Jan Feb	Year] Monthly Operati Useful Power Output (1)	ng Data Primary Energy Source (2)	Supplementary Energy Source (3)	Total Energy Inp (4)
[Prev] Jan Feb Mar	Year] Monthly Operati Useful Power Output (1)	ng Data Primary Energy Source (2)	Supplementary Energy Source (3)	Total Energy Inp (4)
[Prev] Jan Feb Mar Apr	Year] Monthly Operati Useful Power Output (1)	ng Data Primary Energy Source (2)	Supplementary Energy Source (3)	Total Energy Inp (4)
[Prev] Jan Feb Mar Apr May	Year] Monthly Operati Useful Power Output (1)	ng Data Primary Energy Source (2)	Supplementary Energy Source (3)	Total Energy Inp (4)
	Year] Monthly Operati Useful Power Output (1)	ng Data Primary Energy Source (2)	Supplementary Energy Source (3)	Total Energy Inp (4)
Jan Feb Mar Apr May Jun	Year] Monthly Operati Useful Power Output (1)	ng Data Primary Energy Source (2)	Supplementary Energy Source (3)	Total Energy Inp (4)
Jan Feb Mar Apr May Jun Jul Aug	Year] Monthly Operati Useful Power Output (1)	ng Data Primary Energy Source (2)	Supplementary Energy Source (3)	Total Energy Inp (4)
Jan Feb Mar Apr May Jun Jul Aug Sep Oct	Year] Monthly Operati Useful Power Output (1)	ng Data Primary Energy Source (2)	Supplementary Energy Source (3)	Total Energy Inp (4)
Jan Feb Mar Apr May Jun Jul Aug Sep Oct	Year] Monthly Operati Useful Power Output (1)	ng Data Primary Energy Source (2)	Supplementary Energy Source (3)	Total Energy Inp (4)
Jan Feb Mar Apr May Jun Jul Aug	Year] Monthly Operati Useful Power Output (1)	ng Data Primary Energy Source (2)	Supplementary Energy Source (3)	Total Energy Inp (4)

*** End of Exhibit Q-2 ***

Exhibit R-1 Community Interest Requirement Attestation Form

{SCE Comment: The following Appendix R-1 must be provided to SCE before any construction activity at the Project Site.}

[Seller's Letterhead]
[Date]

Southern California Edison Company 2244 Walnut Grove Ave. Rosemead, CA 91770

Attn: Vice President, Power Supply

Subject: CR-RAM Power Purchase Agreement ("PPA") between Southern California Edison Company ("SCE") and [Insert Seller's full legal name], a [insert Seller's form of entity and state of registration] ("Seller") Community Interest Requirement Attestation

Dear Sir or Madam:

Pursuant to section 2.01(d) of the CR-RAM PPA, Seller submits to SCE this Community Interest Requirement attestation. Capitalized terms used but not otherwise defined in this letter have the meanings set forth in the CR-RAM PPA. Seller hereby represents and warrants that:

- 1. This Community Interest Requirement attestation [is] [is not] submitted prior to the start of any construction activity at the Project Site.
- 2. [Seller has obtained:
 - □ at least three (3) separate [Expressions of Interest] [Commitments to Enroll] from three (3) separate CI Eligible Customers; and
 - □ Seller has obtained [expressions of interest in Subscribing to the output of the Project from CI Eligible Customers ("Expressions of Interest") representing at least fifty-one percent (51%) of the Contract Capacity] [commitments to Subscribe to the output of the Project from CI Eligible Customers ("Commitments to Enroll") for at least thirty percent (30%) of the Contract Capacity]. {SCE Comment: Select the applicable type of demonstration. Applicable to Projects that do not have a municipal guarantee of meeting the Community Interest Requirement.}

[Seller has obtained:

□ A guarantee from the municipality where the Project is located that [CI Eligible Customers have committed to Subscribe to at least thirty percent (30%) of the Contract Capacity] [CI Eligible Customers have provided Expressions of Interest sufficient to reach a Subscription rate of fifty-one percent (51%) of the Contract Capacity]. A copy of this guarantee is attached hereto.] {SCE

Comment: Applicable to Projects that have a municipal guarantee of meeting the Community Interest Requirement.}

3. [The CI Eligible Customers who have submitted [Expressions of Interest] [Commitments to Enroll] are as follows:

		Custo	mer Ir	forma	ition		
Customer Name (first and last)	Company/ Institution (if applicable)	Street Address	City	ZIP	County	Service Account (required for Commitments to Enroll)	Subscription (% of Contract Capacity)

{SCE Comment: Applicable to Projects that do not have a municipal guarantee of meeting the Community Interest Requirement.}

4. [Documentation supporting each of the [Expressions of Interest] [Commitments to Enroll] in the form of Exhibit R-2 or R-3, as applicable, are attached hereto.] {SCE Comment: Applicable to Projects that do not have a municipal guarantee of meeting the Community Interest Requirement.}

Very truly yours,	
[Insert Seller's full legal name, form of entity, and	state of registration]
By: Name:	
Title:*** End of Exhi	bit R-1 ***

Exhibit R-2 Expression of Interest Form

{SCE Comment: The top portion of the following form should be populated by Seller with details concerning the Project, and then the lower portion should be completed by each CI Eligible Customer who has expressed interest in Subscribing to the output of the Project. Completed Expression of Interest forms should be submitted to SCE in conjunction with the Community Interest Requirement Attestation (Exhibit R-1) before any construction activity at the Project Site.}

	Project Information
	(to be completed by Seller)
Developer Name	
Developer's Authorized Agent	
Title of Authorized Agent	
Phone Number	
Email Address	
Project Name	
Project Address	
Project Size	
	Customer Information
	y account holder or authorized representative only)
First Name	
Last Name	
Company/Institution	
Phone Number	
Email Address	
Street Address	
City	
ZIP Code	
Proposed Subscription	
Service Account # (not required but	
preferred)	
Customer Account # (not required but	
preferred)	
I,	, confirm that I am the account holder or
authorized representative of the acc	ount holder for the account listed above and am expressing
interest in the project owned by	described
	
	ny current energy service provider, and I acknowledge that
	ice from other energy service providers or Community Choice
Aggregators are ineligible to subscr	ribe to the output of the project described above.
G	D 4
Customer Signature	Date
Authorized Agent Signature	Date
	*** End of Exhibit R-2 ***
The contents of this document are	subject to restrictions on disclosure as set forth in the Agreement.

GTSR – CR Rider Exhibit R-2

Exhibit R-3 Commitment to Enroll Form

{SCE Comment: The top portion of the following form should be populated by Seller with details concerning the Project, and then the lower portion of the Commitment to Enroll form should be completed by each CI Eligible Customer who has committed to Subscribe to the output of the Project. Commitments to Enroll should be submitted to SCE in conjunction with the Community Interest Requirements Attestation (Exhibit R-1) before any construction activity at the Project Site.}

	Project Information
	(To be completed by Seller)
Developer Name	
Developer's Authorized Agent	
Title of Authorized Agent	
Phone Number	
Email Address	
Project Name	
Project Address	
Project Size	
	Customer Information
	y account holder or authorized representative only)
First Name	
Last Name	
Company/Institution	
Phone Number	
Email Address	
Street Address	
City	
ZIP Code	
Subscription Commitment	
Service Account Number	
Customer Account Number	
I,	, confirm that I am the account holder or
authorized representative of the acc	ount holder for the account listed above and am committing
to subscribe to the output of the pro	e e e e e e e e e e e e e e e e e e e
	with the estimated date for when this project will come online,
	g information. I also confirm that SCE is my current energy
	e that customers who receive energy service from other
energy service providers or Commu	unity Choice Aggregators are ineligible to subscribe to the
output of the project described above	ve.
Cystomon Cionotymo	Date
Customer Signature	Date
	<u></u>
Authorized Agent Signature	Date
	*** End of Exhibit R-3 ***

GTSR – CR Rider Exhibit R-3